

# NOTA DI LAVORO

91.2009

The Decision to Migrate and Social Capital: Evidence from Albania

By **Cristina Cattaneo**, Fondazione Eni Enrico Mattei and University of Sussex

# GLOBAL CHALLENGES Series Editor: Gianmarco I.P. Ottaviano

# The Decision to Migrate and Social Capital: Evidence from Albania

By Cristina Cattaneo, Fondazione Eni Enrico Mattei and University of Sussex

# **Summary**

The objective of this paper is to determine whether the participation in social organizations, which are commonly defined as a form of social capital, represents a complement or a substitute with respect to emigration. The nature of the relationship depends on the motivations behind the two choices, which induce the households to join a group and to invest in migration. To address this research question a bivariate probit model is employed, in that the decision to migrate and to join a social organization are estimated simultaneously. Both temporary and permanent emigration of the household are addressed. The results of the empirical estimation reveal that families participating in social organizations are more likely to send siblings abroad permanently, as they may receive from the social network important information that is crucial for permanent emigration. Hence, social capital performs a role as complement to permanent emigration. On the other hand, social capital is associated with a lower probability of moving temporarily. This may indicate that families resort to social capital rather than to temporary circular migration to overcome contingent liquidity constraint and therefore social capital is a substitute for temporary emigration.

Keywords: International Migration, Social Capital, Information Network

JEL Classification: O15

Address for correspondence:

Cristina Cattaneo
Fondazione Eni Enrico Mattei
Corso Magenta 63
20123 Milan
Italy
E-mail: cristina.cattaneo@feem.it

The decision to migrate and social capital: Evidence from Albania

Cristina Cattaneo

Fondazione ENI, Enrico Mattei

and

University of Sussex

December 2008

**Abstract** 

The objective of this paper is to determine whether the participation in social organizations, which are commonly defined as a form of social capital, represents a complement or a substitute with respect to emigration. The nature of the relationship depends on the motivations behind the two choices, which induce the households to join a group and to invest in migration. To address this research question a bivariate probit model is employed, in that the decision to migrate and to join a social organization are estimated simultaneously. Both temporary and permanent emigration of the household are addressed. The results of the empirical estimation reveal that families participating in social organizations are more likely to send siblings abroad permanently, as they may receive from the social network important information that is crucial for permanent emigration. Hence, social capital performs a role as complement to permanent emigration. On the other hand, social capital is associated with a lower probability of moving temporarily. This may indicate that families recourse to social capital rather than to temporary circular migration to overcome contingent liquidity

constraint and therefore social capital is a substitute for temporary emigration.

Keywords: International Migration, Social Capital, Information Network

1

#### 1 Introduction

The fall of the communist regime gave birth to a massive migration movement, which was, to a large extent, the most significant social, political and economic phenomenon experienced by Albania. Initially, this outflow was due to the end of the controls on internal and external migration, although in a more advanced phase, the political instability and the economic downturn gave further impetus to emigration. Italy and Greece are the most important host destinations for Albanian migration. Albania, notwithstanding its effort to build the foundations for a market-based economy, remains one of the poorest countries in Europe. Per capita income is one of the lowest among the transition economies and poverty is pervasive, with 25 percent of the population living in poverty (World Bank, 2007). The poverty experienced by households induced strong migration pressures. The government of Albania estimates that the number of Albanians abroad in 2005 was over one million, representing 30 percent of the total population.

The economic literature has recognized that social network and peer groups play a strong role in challenging poverty and vulnerability, in particular in the context of poorly developed credit and insurance markets. Self-help groups are found to be a crucial source of income for certain vulnerable catergories, such as women (La Ferrara 2002b). Social capital is associated with higher per capita consumption expenditures and a lower probability of poverty (Grootaert, 2000; Grootaert et al., 2002; Narayan and Pritchett, 1997). While the link between social capital and poverty has been investigated empirically, the link between social capital and migration has been relatively under-researched. Given the high rate of emigration and suitable conditions for the generation of social capital, Albania represents an ideal case to study this particular connection.

The research question addressed in this paper concerns the type of relationship existing between social capital and out-migration. In this context social capital is captured by the participation in groups of people who get together regularly either to do an activity or to talk about things. In particular the analysis aims to determine whether the participation in social organizations represents a complement or a substitute with respect to emigration. The nature of the relationship depends on the motivations behind the two choices, which induce households to join a group and to invest in migration. In the case where the two decisions are driven by common underlying factors, it can be expected that social capital and migration act as substitutes for one another.

First, social capital helps reduce risks, as a social network can provide support in case of unanticipated adverse shocks. This is particularly relevant in situations of poorly developed insurance markets, where insurance crucially depends on social connections (Fafchamps and Gubert, 2007a; Fafchamps and Lund, 2003). By the same token, the migration decision can be

generated from a risk-sharing behaviour (Halliday, 2006; 2008; Lucas and Stark, 1985; Rosenzweig and Stark, 1989; Yang and Choi, 2007). If risk reduction is the main engine for both investments, then an individual values two alternative options in order to mitigate eventual risks (i.e., social capital or alternatively migration). Another market failure common in developing countries is represented by poor or non-existing credit markets. In this context, both social capital (Besley, 1995; Grootaert et al., 2002; La Ferrara, 2002b) and migration (Lucas, 1987; Rozelle et al., 1999; Taylor, 1992; Taylor and Wyatt, 1996; Taylor et al., 2003) represent an alternative measure which assists households in coping with financial constraints. This example suits the Albanian context well, as the credit markets suffer from imperfections. For example, only one-third of all farm households are able to access credit from a private bank, if they were to ask for a small loan to start a business, at an interest of about 11 percent (World Bank, 2007).

On the contrary, it can be the case that the decision to participate in social capital activities reflects the need to share information. Social capital often operates through diffusing information regarding opportunities, related to labour and credit markets (Granovetter, 1995). Moreover, the social networks facilitate the flow of information and knowledge between economic agents, whereby actions are stimulated, transaction costs are reduced and additional income is generated (Barr, 2000; Coleman, 1988; Tiepoh et al., 2004). At the same time, migration is a risky investment and its costs can be reduced by means of a network of social connections. The members of the social group in origin countries can act as a channel to spread important information that facilitates migration (Massey and Espinosa, 1997; Palloni et al., 2001; Winters et al., 2001). Under this circumstance, migration and social capital may perform a role of complement for one another.

The migration literature to date is replete with contributions that consider social capital as an instrument for transmitting information that facilitates migration. On the contrary, to my knowledge, there are no empirical contributions that investigate the alternative role of social capital in migration, whereby social connections become a substitute for migration as a risk-coping mechanism.

The remainder of the paper is organized as follows. Section 2 presents a brief review of the literature. Section 3 presents a comparison between temporary, permanent and non-migrant families on the basis of selected characteristics. In Section 4 the individual determinants of group formation are studied, building on the work of Alesina and La Ferrara (2000). Section 5 presents the empirical methodology. Section 6 presents the econometric results and section 7 provides a summary and conclusions.

#### 2 Literature

The research question above implies that a detailed analysis of the determinants of social capital on the one hand and migration, on the other, are defined. While in the economics literature the decision to migrate has been widely analysed, both theoretically and empirically, theoretical models explaining individual motives for the participation in social groups are limited. As is well known, the migration decision exhibits many similarities with an investment decision and, for this reason, original models of migration described migration as an investment that increases the productivity of personal resources (Sjaastad, 1962). Conversely, rather than considering migration as an individual decision, the *New Economics of Labour Migration* (NELM), introduced by Stark and Bloom (1985), emphasized the role played by the family and it diminishes the prominence of migration returns as the sole determinant of the choice. The contributing insight of the authors is that the decision to migrate may take place within a family or a household context, rather than being entirely a decision exercised in isolation by the individuals. Moreover, income gains from migration should be accompanied by other objectives, such as the minimization of risks or the relaxation of constraints in credit and insurance markets. These collateral motivations are interpreted as push factors in migration.

First of all, migration, by means of the remittances inflow, may perform the role of intermediate investment, whereby households can alleviate capital constraints to eventually initiate or enhance self-employment activities. This mechanism is analysed in Lucas (1987) where the emigration is associated in the long-run with increased crop productivity and cattle accumulation. These effects are the consequence of migrant's earnings, which are eventually used in activities not otherwise accessible, such as financing physical investments, new production techniques or for the purposes of insurance. In Rozelle et al. (1999) migrant remittances have a statistical and positive effect on agricultural productivity, as measured by production per unit of land, whereas in Taylor et al. (2003) they enhance cropping income. The existence of such effects can be justified only if capital constraints bind, as under perfect capital markets, the impact should be zero. In Taylor (1992) and Taylor and Wyatt (1996) the authors report that remittances have a more than unitary effect on household-farm income, indicating that they allow for the relaxation of credit constraints on household production.

A second motivation for migration under the NELM framework is that there might be a family risk-sharing behaviour behind the decision, which can induce moving even in the absence of wage differentials. This strategy only requires that earnings at destinations are not correlated or inversely correlated with earnings in origin locations. This is, for example, the result reported in Rosenzweig and Stark (1989), where the households are able to cope with uncertainty and smooth

consumption by means of marriage-cum-migration. Halliday (2006; 2008) finds that the households respond to adverse shocks by re-allocating labour within the family and by increasing the number of male family members living abroad. Remittances from overseas migrants serve as insurance for relatives back home, suggesting that migration is among the mechanisms adopted by households to cope with risks. In Lucas and Stark (1985) and Stark and Lucas (1988) the families most exposed to risks during drought tend to receive greater remittances from urban migrants, which act as insurance during droughts. Yang and Choi (2007) report a considerable response of remittances to income shocks, caused by rainfall. In their paper, a change in household domestic income is negatively associated with a change in remittances, implying that any decline in income, due to rainfall shocks, is replaced by new inflows of remittances to the household.

Within the context of risk diversification, the existence of a social network makes migration a reliable and a relatively risk-free resource, implying that strong ties with current or former migrants influence, *ceteris paribus*, the probability of migration of others within the social network. This hypothesis emphasizes the importance not only of close ties within the families, as predicted by the NELM, but also of diffuse ties within the community, strengthening the validity of the so called social capital theory.

According to this theory, social networks influence the costs and the benefits of migration, and this in turns expands the migration opportunities. In Palloni et al. (2001) the hazard of outmigration is larger among those who have connections with an older migrant, compared to those lacking this source of social capital. The analysis is computed controlling for the influence exerted by conditions which are common among the individuals of the same network. Massey and Espinosa (1997) find that social capital, proxied by friends and relatives with previous migration experience, helps initiate migration between Mexico and the United States and facilitates additional US trips. In their analysis, the strongest role is performed by migration specific social capital, which are connections generated over the course of migration itself. Mora and Taylor (2005) report that the number of family members previously migrated to US is by far the most statistically significant factor influencing migration. Winters et al. (2001) find a positive influence of migrant networks on both the decision to migrate and on the number of migrants that a household sends to US. In McKenzie and Rapoport (2006) community emigration experience, proxing for migration network, influences the impact of wealth on the probability of household head migration. For a high level of network, the budget constraint is less binding and therefore migration possibilities are extended to the less well off-families. To correct for the possible endogeneity of the migration network, the authors introduce demand-side variables of the destinations, as is found a regular pattern of migration trips from Mexican villages to specific US destinations. In Stecklov et al. (2008) whether the role of the network has a gender component is investigated. It is found that while the family network has a positive influence on emigration, regardless of the gender of the mover, the community network is not-gender neutral, in that the community migration experience influences female but not male emigration. Moreover, female migration is largely affected by a network, composed of female family members, with past emigration experience.

Social capital refers to intangible resources available to individuals by means of a network of relationships. The membership in inter-personal networks and social institutions allow individuals to build up other forms of capital, which benefit their position in society. In the various definitions used to explain the meaning of social capital it appears that one of its ultimate results is the mutual benefit of the network members and of the society as a whole. In their seminal contribution, Putnam et al. (1993) show that the different social structure of Italian regions, characterized by horizontal links in the North and hierarchical organizations in the South, has a large effect on the economic performance of the regions. Knack and Keefer (1997) indicate that trust has a positive impact on economic growth, and the scope of the influence is larger, the poorer are the countries, because of their less-developed financial sector, insecure property rights, and the poor enforceability of contracts.

The literature has also emphasized that an important benefit emerging from the existence of social networks is the reduction of risks. Consumption smoothing and risk sharing through informal arrangements with family and friends are among the set of strategies that households employ to cope with risks under weak formal institutions. The literature that analyses the effects of familybased income transfers on consumption smoothing is vast (among others, Deaton, 1992; Gertler and Gruber, 2002; Jalan and Ravallion, 1999; Ligon et al. 2002; Townsend, 1994). However, a growing body of evidence has shown that risk-sharing is not complete within village, but it is limited to the members of one sub-network, such as ethnic groups or family and friends. Fafchamps and Lund (2003), for example, demonstrate the crucial importance of the networks of friends and relatives in the case of unanticipated shock, which provide support by means of gifts and zero interest loans. However, they find that these informal arrangements do not extend to the village as a whole, because of the difficulty for villagers to monitor each other. In Fafchamps and Gubert (2007a) gift and loans serve a risk-sharing purpose among people connected by network ties, but again the households do not engage in links that are optimal from the point of view of income risk-sharing. For this type of network to be optimal implies different income profiles between connected households but this is satisfied only with increasing social and geographical distance. The problem of enforcement and the rising costs of links with distance reduce the chance that such optimal arrangements take place (Fafchamps and Gubert, 2007b).

Another important benefit arising from social capital, which proves crucial in developing countries, is related to the credit transactions, as far as social capital directly increases access to credit. Households participating in local associations are more likely to receive loans and they also receive larger amount of credits, because social capital helps build trust, as suggested by Grootaert et al. (2002). This emerges even if the purpose of the association is beyond financial objectives. La Ferrara (2002b) reports that 64 percent of individuals in a sample of 300 are able to borrow from self-help groups active in local areas in case of need. Ethnic identity sharply influences the individual capacity to borrow from the group through its effect on reciprocity as well as enforcement. This is because within ethnic groups social sanctions are more likely to apply and social norms respected. In this respect, La Ferrara (2003) shows that kinship networks are a valuable mechanism of access to informal credit as the non-anonymity of the dynastic link contributes to the support of the self-enforcing agreement.

#### 3 The data

The data for the analysis are taken from a sample, consisting of a total of 3,840 Albanian households, surveyed in the Albania Living Standard Measurement Survey (ALSMS) 2005, by the World Bank and Albania Institute of Statistics. This round of the LSMS was conducted in the field between May and July 2005 and additional visits to agricultural households followed in October. The survey gathers information at individual, household as well as community levels, which include among other things, modules on migration, fertility and social capital.

The sample for this analysis includes 3,094 families, located both in rural and urban areas, and it includes only the households whose heads are of working age. International migration, as well as internal migration, can be classified according to different criteria, namely whether it is a temporary or a permanent phenomenon. Permanent migration defines a move which lasts for a long period, and which may not imply a return to the place of origin. Conversely, temporary migration occurs if an individual moves abroad and intends to return home after a short period of time. The literature shows that the motivations behind returning can be found, among others, in stronger preferences for consumption at home than abroad (Djajic and. Milbourne, 1988), in higher purchasing power in the origin than in destination countries (Dustmann, 1997; Stark et al.1997) or in specific location preferences (Dustmann, 2001; Hill, 1987). Yang (2006) distinguishes between life-cycle and target-earning type of considerations for explaining return migration. This study supports the life-cycle approach, in that improved economic conditions in destination countries reduces migrants returns, although it does not neglect the possibility that some migrants are motivated by target- earning considerations.

The permanent migration status in the ALSMS 2005 is captured through a fertility module, which records the full list of children, who no longer are resident with the family because of migration. In this respect here are classified as permanent migration households, the families who have at least one child who migrated abroad and who is no longer a member of the household. On the contrary, temporary migration households are those whose heads spent at least one month in a foreign country and returned to the family of origin. Clearly, the proxy used for permanent migration has some limitations as it assumes that the children will permanently stay abroad given that they are no longer part of the family, and not through information on time abroad or the reported intention of the movers. Unfortunately, there is no way to capture the future intentions of those individuals who were abroad at the time of the interview. It should be noted that the approach used here to classify permanent migration has been used extensively in the literature (Carletto et al., 2006; Pinger, 2007; World Bank, 2007). Finally only the most recent episodes of migration are considered here and in both cases, an individual is a migrant if the move occurred within the last year.

Table 1 presents a comparison between temporary, permanent and non-migrant families on selected characteristics, providing the proportion of households belonging to the different categories. In 2004, five percent of the families in the sample invested in temporary migration and four percent in permanent migration, with no overlap between the two migration groups. Overall 279 families, corresponding to nine percent of the sample, send at least one family member abroad, either temporarily or permanently. These figures suggest that emigration in Albania is still a substantial phenomenon. In spite of the mass emigration that occurred after 1990, and continued till the end of the 1990s, emigration continues even in the face of improved economic conditions. It should be noted that the above rates are based on household-level data, and therefore disguise the effect of multiple migrants per household. This implies that the emigration rates for the population as a whole might be notably larger.

The temporary migration households are mostly headed by males and this draws a significant distinction between the gender attribute of the temporary migration families and the non-migration households. This conclusion is suggested by the z-score for the difference in proportions reported in Table A1. On the contrary, 11 percent of permanent migration households have a female head and this latter proportion is in line with the non-migration households (Table A2).

The temporary migrant families are overall less educated than non-migrant ones, in that the migration families, compared to the non-migrant ones, are largely represented in the primary

<sup>&</sup>lt;sup>1</sup> This is done in agreement with McKenzie and Rapoport (2007), where the migrant households are defined through the emigration of the household head.

education group and are less likely to be in the university category, with a statistically significant difference (Table A1). A similar picture emerges for the permanent migration families as, compared to the non-migrant counterparts, are more represented in the primary education group and less represented in the secondary and vocational categories, and the difference in proportions is statistically significant (Table A2). In terms of educational attainment, the heads of temporary and permanent migration families display respectively one and two less years of schooling compared to non-mover households.

#### TABLE 1: HERE

The literature traditionally predicts that migrants are highly educated, given that schooling plays an important role in alleviating the costs and the risks of migration. It is also true, however, that the educational gap between migrants and non-migrants is influenced by the geographical proximity of the home with the destination nation. The closer the destination, the smaller are the emigration costs and the less emigrants are positively selected in terms of education level. For example, the literature on Mexican-US migration finds that Mexican migrants are not endowed with particularly high education levels (Hatton and Williamson, 2004; Mora and Taylor, 2005; Stark and Taylor, 1991). The Albanian case may corroborate this stylized fact, as temporary movers, who mainly chose the nearby destinations of Italy and Greece, are endowed with less than average years of education.

The distinction between migration and non-migration households appears also in terms of location. Families with temporary movers are more likely to belong to the central and mountain areas of Albania compared to non-migration ones. On the contrary, households in the permanent migration group are disproportionately located in the coastal area. The literature suggests that the characteristics of sending communities influence the duration of stay, with migrants from communities with better economic opportunities remaining longer in the destinations (Reyes, 2001). This result is in agreement with the present analysis, in that Tirana and the coastal districts are the regions registering the higher per capita consumption levels (World Bank, 2007) and the larger permanent migration. Migration tends to be a rural phenomenon, regardless of the type of migration. Both temporary and permanent migration household are significantly more likely to be drawn from rural areas than non-mover families.

The data also allow on the extension of the summary analysis, given the availability of additional information related to temporary return migrants. Table 2 shows the responses on selected variables, which describe the motivation and typology of stay abroad for temporary moves. First, it is clear that temporary migrations have a very short duration, with nearly 70 percent of the respondents staying in destination countries for less than six months. Nearly all migrants who have

already been abroad previously plan to move again in the future for a short period of time. The occasional nature of temporary migration can be further inferred from the fact that only in a limited number of cases, the spouse and eventually the children follow the head of the household to the destination countries. In the literature this type of migration is defined as circular in that migrant workers move frequently between the host and the source country and they stay at destination only for a short period, such as the harvest season (Dustmann and Weiss, 2007). The reason for returning to Albania eventually results from an ex-ante intention. Many individuals move to perform a prearranged a seasonal job, which has a limited duration, or they returned for family reasons. Only in few cases did an unsuccessful experience abroad motivate the return (no residence, no work or expulsion). Finally, the individual network which provides help for migration consists of contacts, family or friends at destination, rather than in Albania.

The final issue regards the link between social capital and migration. It should be noted that the measurement of social capital for empirical use has encountered several concerns given the difficulties in finding an appropriate proxy for it. A well applied strategy is to focus on one of its component, namely the participation in group associations (Alesina and La Ferrara, 2000; Costa and Kahn, 2003; La Ferrara 2002a; Glaeser et al. 2000; Glaeser et al. 2002). As Putnam suggests, involvement in social groups and associations is conducive to generating the beneficial effects of social capital, and therefore social capital and group membership are likely to be highly correlated. The participation in associations comprises the membership in groups of people who get together regularly to do an activity or talk about things. The household is defined as a member of a group if the head of the household declares an attachment to one or more associations of the following type: labour related, village, cultural, religious, environmental, youth, veterans, sport, ethnic or other associations.

#### TABLE 2: HERE

Participation rate in Albania is quite low. Only 676 households out of 3,094 - corresponding to 22 percent of the families - report membership of at least one association. In the whole sample, the average number of group membership is 0.3 per household. The most important associations are labour type groups, which include farmer, irrigation activities, traders or business, professional and trade unions. About 36 percent of the household indicate these as being the most relevant groups. Political associations are indicated as the most relevant by 15 percent of the

\_

<sup>&</sup>lt;sup>2</sup> The variable proxing for social capital in this analysis is different from what is typically used in the empirical literature that focuses on social capital in a migration context. In the latter, the social network has a strong migration component, being captured by past emigration experience of family and friends. Here, however, given the larger role that is given to social capital, as it is assumed that social capital not only helps proving information for emigration but it serves as risk coping mechanism, a more general definition is applied.

families, village type groups by 11 percent and finally religious groups by eight percent (see Table 3). Very few pay fees to the organizations. About a fifth of members of religious and political organizations contribute with money, and less than ten percent pay in the other organizations.

Placed in a comparative setting, the participation rates in Albania are lower than in other countries, both developed and developing. For example, Beugelsdijk and van Schaik (2005) report that the average group participation within seven western European countries is 62 percent for passive membership, and 41 percent for active membership. Active membership requires not only membership but also active voluntary work for the association. In US, Alesina and La Ferrara (2000) and Glaeser et al (2002) quote that at an individual level, 71 percent of respondents participate in at least one group, and the average group membership is 1.8 per person. In South Africa group participation involves 71 percent of the households, with an average number of membership in 1.3 groups (Maluccio et al., 2000). In Tanzania La Ferrara (2002a) report that 72 percent of individuals are members of some groups, and the average number of group membership is 1.6 per person.

#### TABLE 3: HERE

A possible explanation for such low participation in Albania is that during transition many networks of agricultural and industrial cooperatives and work units, which were active during the communist period, were closed or disintegrated (World Bank, 2002). After the shutting down however no other economic organizations were developed to replace the old ones. Other types of non-production, non-business associations such as religious organizations, charity groups, self-help organizations, and public interest groups have been developed, but they lack mission and participation. Moreover, Albanians demonstrate a sceptical feeling towards agricultural and industrial associations, which remind them of the socialist cooperatives from older times.

The preliminary analysis of social capital and migration in Table 4 reveals that families participating in social organizations are poorly represented in the temporary migration category. The second column of the table reveals that on the whole sample, five percent of the households engage in temporary migration. However, if we consider only the households participating in associations, the proportion reduces to three percent and it reaches six percent among the non-participating ones. Moreover, the difference in proportions between participating and non-participating groups is statistically significant. This may provide preliminary evidence that the participation in social groups performs the role of substitute for temporary emigration, as social capital satisfies the same objectives that motivate emigration. In contrast, the permanent migration

households appear to be equally represented in the different sub-samples, as indicated in the third column of the same table.

#### TABLE 4: HERE

# 4 Some background theory on social capital

In analysing the effect of social capital on a particular outcome there is a fundamental challenge determined by the fact that individuals choose the persons they want to be friends with as well as of the groups they want to be a member. It is argued that people select as friends those persons that are similar to them (Mouw, 2006). Moreover, it is possible that much of the estimated effect of social capital is generated by the fact that "common social capital grouping are subject to common human capital influence" (Palloni et al., 2001). The kinship or friendship links underline the existence of shared common characteristics, which induce similar behaviour among the individuals within the group. In this respect, the casual effect of social capital on a particular outcome may simply be the spurious result of a correlation between unobservable common features which, in the first instance, determine participation in a specific group and influence the behaviour within the group. Therefore, to avoid the introduction of biases in estimation, one should net out the effect of common human capital influences that exist within the network. Social capital is thus treated as "an endogenous outcome of decisions that are contemporary to the behavioural migration choice" (Durlauf, 2002), and estimation requires the use of instrumental variable techniques.

In order to uncover adequate identifying instruments for estimation, a theory which explains the observed differences in the social capital among the individuals is helpful. In this regard, the theoretical model developed by Alesina and La Ferrara (2000) is employed here, where the heterogeneity of the population is identified as one of the determinants of group participation.

The model considers a community where only two types of individuals live: individual type B and type W. The utility from group participation is influenced by individual characteristics along with the composition of the group:

$$\mathbf{u}_{i} = \mathbf{u} \left( \alpha_{i}^{B}, \mathbf{P}_{W} \right) \text{ if } i \in \mathbf{B} \tag{1}$$

$$\mathbf{u}_{i} = \mathbf{u} \left( \boldsymbol{\alpha}_{i}^{W}, \mathbf{P}_{B} \right) \text{ if } i \in \mathbf{W}$$

$$u_{\alpha}(.) < 0$$
,  $u_{p}(.) < 0$ 

where  $P_B$  ( $P_W$ ) denotes the proportion of type B (W) participating in social groups, while the parameter  $\alpha$  identifies individual preference toward participating in a group. The model assumes a preference toward homogeneity and that a higher value of the parameter  $\alpha$  denotes a lower interest in group participation.

Given the reservation utility  $\overline{u}$ , individual i of type B and of type W choose to participate if respectively:

$$u(\alpha_i^B, P_W) \ge \overline{u}$$
 and  $u(\alpha_i^W, P_B) \ge \overline{u}$ 

which implies that:

$$\alpha_i^B \leq g(\overline{u}; P_W)$$
 and  $\alpha_i^W \leq g(\overline{u}; P_B)$ 

Given the same cumulative distribution of  $\alpha_i^B$  and  $\alpha_i^W$ , denoted by F(.), the total mass of individuals of type B and W who chose to participate in social activities is represented by:

$$\widetilde{B} = F(g(\overline{u}, P_{W})) * B$$
(3)

$$\widetilde{W} = F(g(\overline{u}, P_R))^* W \tag{4}$$

And therefore, the aggregate level of participation is:

$$S = \frac{\widetilde{B} + \widetilde{W}}{B + W} \tag{5}$$

An equilibrium is a group composition  $(P_B^*, P_W^*)$ . The equilibrium condition is defined by a situation where for both types, none of the group members wish to leave and none of the non-members wish to join. In equilibrium, the proportion of type B in the group  $(P_B)$  must be given by:

$$P_{\mathbf{B}} = \frac{\widetilde{B}}{\widetilde{B} + \widetilde{W}} \tag{6}$$

$$P_{\mathbf{W}} = 1 - P_{\mathbf{B}} \tag{7}$$

The aggregate level of participation S is the share of the total population who participates to a group. Combining equation (3.5) with (3.3) (3.4) (3.7) and then setting  $\omega = \frac{W}{W+B}$ , it follows that:

$$S = F(g(1 - P_R^*))^* (1 - \omega) + F(g(P_R^*))^* \omega$$
(8)

Notice that  $\omega$  represents the degree of heterogeneity of the population.

Equation (8) states that group participation is influenced by the degree of heterogeneity in the society, with higher fragmentation implying lower participation. The authors demonstrate that a shift toward a more heterogeneous society, in which the fraction of the most abundant type W decreases and the fraction of the minority type B increases, induces a loss in participants of the type W, which is not compensated by an increase in participation of type B, thereby the total overall participation reduces.

This conclusion is supported by Knack and Keefer (1997) who state that social capital is undermined by heterogeneity. The greater the distance in preferences embedded in a polarized society, the greater the increase in the probability of unstable policy coalitions and this negatively affects trust among individuals. Moreover, on the one hand, polarized societies lack common

backgrounds among individuals, and this hinders self-enforcing agreements, and on the other, they are characterised by rent-seeking behaviour, which reduces trust.

For empirical purposes, to proxy for the population heterogeneity, the literature widely uses an index of ethno-linguistic fragmentation (ELF), which is computed as follows:

Fragmentation<sub>i</sub>= 
$$1 - \sum_{i} p_{ji}^{2}$$
 (9)

where i indicates the geographic unit of observation such as districts, j the different ethnicity groups and  $p_{ji}$  is the proportion of ethnicity j in the population of district i. Clearly, ethnicity is only one of the possible dimensions along with the fragmentation index can be computed, as the population may differ with respect to other characteristics, such as religion, economic activity or education. As it will be discussed later, in this analysis only the fragmentation in ethnicity and in economic activity are used.

# 6 Econometric and Methodology

To estimate the effect of social capital on out-migration, an impact dummy that captures the participation in a social organization is introduced in a probit migration function. However, to control for the existence of the unobservable heterogeneity described above, which may bias the estimated parameters of a single equation probit model, a structural approach is adopted, employing a recursive model, which treats social capital as an endogenous regressor. The recursive structure is modelled by a reduced form equation of the potentially endogenous variable (i.e., social capital) and a structural form equation that defines the outcome of interest.

Let  $y_1^*$  and  $y_2^*$  be, respectively, household's unobservable propensity to migrate and the household's unobservable propensity to participate in group associations.

$$y_1^* = \mathbf{x}_1' \mathbf{\beta}_1 + \mathbf{u}_1 = \delta_2 y_2 + \mathbf{z}_1' \mathbf{\delta}_1 + \mathbf{u}_1$$
 (10)

$$\mathbf{y}_{2}^{*} = \mathbf{x}_{2}^{'} \mathbf{\beta}_{2} + \mathbf{u}_{2} \tag{11}$$

y<sub>1</sub> and y<sub>2</sub> are dichotomous variables observed according to the rule:

$$y_j = 1 \text{ if } y_j^* > 0 \text{ and } y_j = 0 \text{ if } y_j^* \le 0 \text{ j=1, 2}$$

-

<sup>&</sup>lt;sup>3</sup> The index of heterogeneity measures the probability that two individuals, randomly drawn from a district, belong to different ethnicities. This index has been used among others by Alesina et al. (1999); Alesina and La Ferrara (2000); Easterly and Levine (1997) and La Ferrara (2002a).

In particular,  $y_1 = 1$  captures the extent of migration and  $y_2 = 1$  captures group participation. Consistent estimators of this model could be produced applying a two-steps procedure, where two probit functions are sequentially estimated and the first stage participation predictions are used in estimating the second stage migration function. However, this methodology has the potential pitfall that it fails to account for possible correlations between the disturbances in the two equations and for this reason it is potentially inefficient (Greene, 1998). A consistent estimator (and one that is fully efficient) is represented by the bivariate probit model described in Greene (2000).

The key reason for the use of the bivariate probit in this application relates to the notion that both outcomes being modelled are potentially jointly determined, and/or the unobservables influencing the outcomes are simultaneously correlated through some process or other. Therefore, we assume that  $u_1$  and  $u_2$  are bivariate normal, with zero means, unit variance and covariance  $\rho$ . Given the possibility of a non-zero covariance across the error terms, the system of equation is also referred to as a seemingly unrelated regression model (SURE). Despite the two equations in the system contain their own vector of coefficients ( $\beta_1$ ,  $\beta_2$ ) and for this reason may appear unrelated, the link in the error terms between the two equations is exploited. The estimated  $\rho$  captures the relationship between the unobservables governing the two decisions. The test on  $\rho=0$  is interpreteable as a test on the exogeneity of the social capital variable and in the situation of  $\rho=0$  the model collapses into two separate independent probit models.

The maximum Likelihood function of the simultaneous model is expressed as:

$$L = \prod_{y_{1}=1, y_{2}=1} G(\mathbf{x}_{1}'\boldsymbol{\beta}_{1}, \mathbf{x}_{2}'\boldsymbol{\beta}_{2}; \rho) \prod_{y_{1}=1, y_{2}=0} G(\mathbf{x}_{1}'\boldsymbol{\beta}_{1}, -\mathbf{x}_{2}'\boldsymbol{\beta}_{2}; -\rho) \prod_{y_{1}=0, y_{2}=1} G(-\mathbf{x}_{1}'\boldsymbol{\beta}_{1}, \mathbf{x}_{2}'\boldsymbol{\beta}_{2}; -\rho) *$$

$$* \prod_{y_{1}=0, y_{2}=0} G(-\mathbf{x}_{1}'\boldsymbol{\beta}_{1}, -\mathbf{x}_{2}'\boldsymbol{\beta}_{2}; \rho)$$
(12)

To ensure identification, a set of instrumental variables are required. These should affect the group participation  $(y_2)$ , but should be orthogonal to the error process in the migration equation  $(y_1)$ . For this purpose, building on the theoretical model outlined, an index of fragmentation is computed, capturing the degree of heterogeneity in Albanian society.

# 7 Empirical Evidence

# 7.1 Single Probit Estimation

In this analysis, consistently with the NELM models in migration, the family, rather than the single individual, becomes the unit of analysis, given that household members act collectively to

maximize utility.<sup>4</sup> Migration is thus considered part of a household strategy to minimize risks or overcome liquidity constraints in credit and insurance markets. To capture both temporary and permanent emigration experience, two alternative dependent variables are used in formulating the migration function.

The set of explanatory variables (X<sub>1</sub>) includes the demographic characteristics of the household, captured by the age and the gender of the household head.<sup>5</sup> The gender of the household head should influence the decision to migrate, as female headed households may be more vulnerable to risks and therefore recourse to migrate more often. Second, to proxy for the human capital potential of the family, the education of the household head is used. The benefits and costs of migration are influenced by human capital, as the returns to migration and the costs and risks of emigration select the individuals in terms of skill levels. For example, low rewards to education in destination markets together with limited risks of emigration because of near-by destinations, are found to be responsible for a low propensity to migrate among the highly educated. Third, to control for unobservable geographical fixed effects, location dummies are introduced along with a dummy variable for urban households. The key social capital variable, indicating whether the household participates in group associations is introduced.

Two measures of household assets are introduced, namely an agriculture landholding variable and an index that records the ownership status of different non productive assets. Given that the sample is composed by both rural and urban households, the landholding represents a good proxy for wealth in rural areas, whereas the non-productive index is better designed for the urban families. These variables measure the household income generation potential and the ability to secure against risk. On the one hand, wealthier households are less exposed to risks and therefore they should have a lower propensity to migrate. On the other, poorer families may lack resources to finance migration, and therefore a greater capital should enlarge emigration propensity.

The non-productive index is computed as follows. Given a vector of variables capturing wealth,  $W=(W_1,\,W_2,\,...,\,W_j$ ), the index for the single household (I<sub>i</sub>) is computed applying the principal component technique:

<sup>&</sup>lt;sup>4</sup> The NELM models were originally developed in a context of rural-urban migration. However, the extension to international migration from developing to developed countries is straightforward given the high income differentials and the low wage correlations between the source country and the foreign destinations (Taylor and Martin, 2001).

<sup>&</sup>lt;sup>5</sup> The household size is often used as additional demographic factor affecting migration. However, in this study it does not influence emigration propensity and for this reason it was removed.

<sup>&</sup>lt;sup>6</sup> The choice of the non-productive asset to control for wealth in the migration equation is done in line with Rozelle et al. (1999) and Taylor et al. (2003). Alternatively, the household total consumption can be used, as it is considered a good proxy for household life-time resources. However, in this study, the consumption variable does not exert a statistically effect on household migration and for this reason it was not used.

$$I_{i} = a_{1} * \left(\frac{W_{i1} - W_{1}}{s_{1}}\right) + a_{2} * \left(\frac{W_{i2} - W_{2}}{s_{2}}\right) + ... + a_{J} * \left(\frac{W_{iJ} - W_{J}}{s_{J}}\right)$$

where  $a_j$  is the *first principal component* for the j th asset,  $W_{ij}$  indicates whether the household i owns the specific asset j and  $W_j$  and  $s_j$  represent, respectively, the mean and the standard deviation of the ownership of asset j among the households. Computationally, the principal components are assigned so that the maximum discrimination in asset ownership is provided. In this way, the decomposition of the covariance matrix of the asset variables allows us to define a series of uncorrelated linear combinations of the variables that contain most of the variance. The advantage of this index is that it aggregates into a single measure a range of different variables, which individually may not be sufficient to differentiate the welfare characteristics of the household. In this study, the assets and dwelling characteristics that enter the index are: the area of the dwelling, water and sanitary facilities of the house, the possession of coloured television, phone and terrace. Table A3 provides a summary output of the variables.

The asset index could be endogenous to the migration decision, as it may be influenced by migrant remittances. For example, the assets that enter the index may be purchased through remittances. To limit the potential endogeneity, the index is computed considering the dwelling condition and the durable ownership in 1990.<sup>7</sup>

The land variable may also be endogenous, as remittances from past migration may be used to buy land. It should be noted however that the land purchase activity in Albania is underdeveloped, despite the right of private property is guaranteed and the transactions are allowed by law. Different factors are responsible for such low development. The land owners are now entitled to land property rights but after 50-years of collective property, are reluctant to sell land. The process of registration is constrained by conflicts between past and present owners, irregular title deeds, high costs of transactions and by the introduction of regulation on the documentation required for the transaction of land (Sabates-Wheeler and Waite, 2003). Consequently to these inefficiencies, the modification in the structure of land and real estate ownership is limited, and this significantly reduces the potential endogeneity of the land variable.

Column (1) of Tables 5 and 6 reports the results of the single equation probit model, based on the two different migration measures. Table A5 in the Appendix reports the marginal effects. Overall, the estimated coefficients have the expected sign and the regression models display a reasonable fit to the data, as indicated by the Pseudo R<sup>2</sup> measures. The age of the head of the

<sup>&</sup>lt;sup>7</sup> The correlation coefficient between the asset index at the time of the survey and in 1990 is 0.60. This indicates that eventual asset depletions that occurred between 1990 and 2005 did not alter the relative position in the welfare distribution of the households.

household dampens the temporary migration incentive, as suggested by the negative and statistically significant coefficient of the age variable in Table 5, while it increases permanent emigration of the siblings (Table 6). This may indicate that the role of the migrant is assigned to the head when the family is young and then it transfers to the children when the head ages.

#### TABLE 5: HERE

Temporary migration is significantly influenced by the gender of the head. In particular, being a male head increases the probability of temporary migration by one percentage point. This result may suggest that female heads are not in the condition to migrate. In this sample, 77 percent of female heads are either divorced or widow, whereas among male heads the proportion falls to one percent. This eventually implies that the duties connected to the status of single parent reduce the chance to move abroad, even if these families are likely to be the most at need to move. The gender of the household head, on the contrary, has no influence on permanent migration of the siblings as indicated by the non-significant effect of the male dummy. Education is not related to migration decision, as suggested by the negative but non-significant coefficient of the human capital variable. Location marginally influences the likelihood of temporary migration, in that living in Tirana compared to the mountains reduces emigration by 0.7 of a percentage point.

An interesting feature is that wealthier households are less likely to migrate, both permanently and temporarily, as indicated by the negative and statistically significant coefficient of the non-productive asset variable, even if the magnitude of the effect appears negligible. A one unit increase in the wealth index, which is expressed in standard deviation, reduces the probability of temporary migration and permanent migration respectively by 0.2 and 0.6 of a percentage point, on average and *ceteris paribus*. This may indicate that well-off families have less need of capital through migration, and only those families living in poverty recourse to migration as a means to escape from it. Vullnetari (2007) reports that emigration for Albanians represents a survival strategy. The shortages of job opportunities during the 80s, the high inflation and unemployment, which followed the ultra-liberal reforms in the 90s, and the loss of savings, houses and properties after the collapse of the pyramid scheme in 1997, produced a mass emigration, which became the only means of survival for many households. Albanians consider migration "as the most effective way of coping with the country's disastrous economic conditions" (King, 2005). Other studies document a negative link between emigration and welfare, corroborating the idea that migration helps mitigate the lack of financial resources (Rozelle et al., 1999; Taylor et al., 2003).

Controlling for wealth, the ownership of land positively influences the probability of both types of migration. This may indicate that land ownership increases the need for capital, to engage

in productive activities in rural areas. The existence of imperfections in land and capital markets renders households with more agriculture landholding more capital constrained. Land, on the one hand, creates demand for complementary investments, but on the other does not represent a direct source of capital to secure against risk. The positive effect of land on emigration is also found in Massey and Espinosa (1997), Rozelle et al. (1999), Taylor and Wyatt (1996) and Taylor et al. (2003).

Finally, the estimated coefficient of social capital reveals that the participation in social organizations reduces the probability of temporary migration by 0.4 of a percentage point, on average and *ceteris paribus*. In contrast, social capital does not influence the permanent migration of siblings. This result may tentatively suggest that social capital substitutes for temporary migration, in that the household may benefit from social relationships as a source of finance in the case of need. It seems that sibling migration and social membership may respond to different strategies as indicated by the statistically insignificant impact of social capital on migration for this group.

#### TABLE 6: HERE

As mentioned above, however, the single equation probit model fails to control for the correlation in unobservables, influencing both the participation in a specific group and the behaviour within the group. This omission could potentially bias the estimated parameters in this probit model. It is to addressing this particular issue that attention now turns.

#### 7.2 Bivariate Probit Estimation

A structural, recursive model is now estimated, where the determinants of migration are analysed along with the determinants of group participation, in accordance with the system of equations presented in (10) and (11). The vector  $(X_2)$  of explanatory variables for social capital includes the same set of variables used in the migration equation, since it is assumed here that the two decisions are taken simultaneously at the household level. The factors that affect migration are therefore also supposed to influence the decision to participate in group associations. Moreover, in agreement with the theoretical literature on social capital (Alesina and La Ferrara, 2000; La Ferrara 2002a), a measure of heterogeneity in the population is used in the social capital function, providing the role of identifying instrument.

The population heterogeneity can be captured by the index of fragmentation, described above and computed for various dimensions, such as ethnicity, religion, economic activity or

education. In this analysis however only the fragmentation in ethnicity and economic activity are used, given that heterogeneity in education and in religion failed to provide good instruments. For heterogeneity in ethnicity, *j* represents the following races: Albanian, Greek, Roma, Macedonian, Montenegrin, Vllahe and other. For heterogeneity in activity, *j* represents self employed, worker in a household farm, employed in government, public sector, army or state owned enterprise, employed in a private enterprise or other private company, and unemployed. The geographic units of observation for the computation of the index are the 36 Albanian districts.

As reported in Table A4, the level of activity heterogeneity in the population is quite high, indicating a 80 percent chance that two individuals in the same district are employed in different activities. On the contrary the average index of ethnic fragmentation is very low, suggesting little evidence of ethnic heterogeneity in Albania. This is not surprising as 98 percent of the population belongs to the Albanian majority ethnic group. For this reason, the preferred measure of heterogeneity is represented by the activity index and the ethnicity index is included only to provide a robustness check. In addition, it can be argued that the higher the ethnic fragmentation in a country, the higher the level of ethnic tension, which then influences a predisposition towards emigration. According to this notion, the index of ethnic fragmentation may fail to exert the role of instrument as it may be correlated with the probability of emigration. However, given that ethnic fragmentation in Albania is very limited, this concern should not represent an issue for this analysis.

The results of the structural model are presented in columns (2) to column (7) of Tables 5 and 6. The explanatory variables in the social capital equation overall exert well determined effects, which are robust in all the specifications. The coefficient of age is positive, indicating that participation monotonically intensifies with increasing age. This may indicate that siblings of young age increase family duties and reduce the time available for social activities in "young families". This result is not entirely in agreement with existing studies in that social capital is found to increase with age but then decline. For example, Glaeser et al. (2002) find that social capital displays an inverted U-shape pattern over the life-cycle, with a trough in participation between 40 and 49 years of age. This is interpreted as organization membership being a proxy for the stock of

\_

<sup>&</sup>lt;sup>8</sup> The education fragmentation index resulted not correlated with group participation while the religious heterogeneity was correlated with emigration probabilities and therefore it was not orthogonal to the error process in the migration equation.

<sup>&</sup>lt;sup>9</sup> The theoretical model predicts that heterogeneity negatively influences group participation. Given the low level of ethnic heterogeneity, one may expect very high participation rates in Albania. However, as already noted, this is not the case, as the participation rates in Albania are very low. To reconcile the theoretical prediction and the empirical evidence, one may consider the peculiar economic background of Albania. The sceptical feeling of Albanians towards group associations, developed after the collapse of the communist regime, may constitute a strong determinant that adds to ethnic fragmentation in explaining group participation.

social capital, which depreciates among the elderly. To determine the existence of such effects, the square of age was introduced, but the quadratic effect was not found to be statistically significant.

Participation in female headed households is lower compared to the male headed ones, as women may carry the burden of both job and household care and thus have less time to engage in social organizations. Moreover, women are probably more likely to exploit informal social networks, rather than the formal types. Education increases participation, with the coefficient of years of completed schooling being positive and statistically significant. The same effect is found in Alesina and La Ferrara (2000), Glaeser et al. (2000), Glaeser et al (2002) and La Ferrara (2002a). This is consistent with the idea that more educated people have higher confidence in their ability to influence social political decisions and for this reason they are more likely to be involved in social interactions. In addition, this result emphasizes the possibility that schools are an important place for teaching social skills as well as that larger benefits from social interactions result with a higher level of human capital. Urban families are more likely to invest in social capital, suggesting that social meeting are more easily performed in cities rather than in remote rural areas. Urban public transportation connections may facilitate interactions and increase the variety of organizations, which may better suit individual preferences.

The negative relationship between wealth and social capital may indicate that, when financial markets are imperfect, the households eventually recourse to social capital to overcome liquidity constraints. Richer households are not financially constrained and do not need to employ this strategy in case of need. Moreover, Glaeser et al. (2002) suggest that for higher opportunity costs of time, the level of social capital might decline, and this produces a negative association between income and social capital. Nevertheless, their empirical evidence does not support this view, as the estimated coefficient of household income yields a positive sign in the analysis. The positive link between the probability of being member of a group and household income is also reported in Alesina and La Ferrara (2000) and Glaeser et al. (2000), indicating that in their analysis social capital is a normal good. In La Ferrara (2002a) the relationship between wealth, proxied by a non-productive asset index, and social capital not only is increasing but is also concave, as suggested by the positive coefficient of wealth and the negative coefficient of its square. Finally, the coefficients of land and its square indicate that families owning plots of agriculture land are more likely to join social groups and that the relationship is concave.

The estimated coefficient of the activity fragmentation, reported in columns 2 of Tables 5 and 3.6 is negative. As anticipated and in agreement with the theoretical background, the probability of participating in social groups reduces with more fragmented communities, even if the coefficient is marginally insignificant in the temporary migration specification. As a robustness

check, in columns 4, activity fragmentation is replaced by ethnicity fragmentation. Not only do the heterogeneity coefficients remain negative and well determined in all specifications, but the coefficients of the other covariates are not affected by the use of the alternative identifying instruments. Finally, when both measures are introduced, their coefficients remain negative and there is little material change to the estimated effects of interest.

As already discussed, the ethnicity fragmentation may not be independent of the migration decision, as ethnic sorting as well as ethnic tension may represent motivations for emigration. To provide evidence of the validity of both instruments, the fragmentation indices are introduced into a migration probit model. The tests on the significance of the coefficients may corroborate their usefulness in the current application. As indicated by the orthogonality test, provided in the Tables 5 and 6, the instruments are orthogonal to the error process in the migration equations, providing support to the identification of the rho term.

Turning to the migration estimates, the effect of the covariates can be evaluated in terms of their marginal effects. It should be noted that three types of marginal effects can be computed in a bivariate probit model, namely the marginal effect based on the marginal probability, the marginal effect based on the joint probability and the marginal effect based on the conditional probability. In this analysis, the first represents the marginal effect of the covariates on the unconditional probability of migrating. The second is the effect on the joint probabilities of migrating and participating in a group, while the third is the effect on the probability of migrating, conditional on participating in social groups. Appendix B provides a detailed description of the different formulae.

For comparison purposes with the single probit estimation, only the marginal effects based on the marginal probabilities will be discussed. The computed marginal effects are reported in Table A5. For some variables a substantial variation occurs compared to the single probit estimation. The coefficients of the education variable are now negative and statistically significant, both in the temporary and in the permanent migration specifications, suggesting that human capital is not a critical asset for emigration in the Albanian context. The marginal effect based on the marginal probabilities indicates that an additional year of schooling of the head is associated with respectively 0.1 and 0.2 percentage point decrease in temporary and permanent emigration. This is a common finding when migrants face low returns to migration in a foreign market and when international destinations are at short distance (Hatton and Williamson, 2004). This feature characterises emigration from Albania and is also well documented for Mexico-US migration (Mora and Taylor, 2005; Taylor, 1987; Winters et al., 2001). Even if migrants from Albania used to be endowed with higher than average education, in more recent years this feature reversed, as a larger number of less educated individuals migrated. The deterioration of the level of education among

migrants is due to the fact that older cohorts, endowed with less education, add to the mass of movers, as a consequence of economic hardship (World Bank, 2007). The present study, which considers the most recent wave of emigration, confirms this finding.

Compared to the mountain regions, temporary emigration is less likely to occur from coastal areas, while permanent emigration from central regions. Families from coastal and central areas are respectively 0.7 percentage point more likely to migrate temporarily and 1.5 percentage point more likely to migrate permanently, on average and *ceteris paribus*.

Finally, the interpretation of the estimated correlation coefficient  $\rho$  has importance in that it represents a test on the exogeneity of the social capital variable. If  $\rho$  is not statistically different from zero, social capital is assumed exogenous and the model collapses into two separate independent probit models. The Wald tests on the significance of  $\rho$  dictates the choice of the appropriate methodology for temporary and permanent migration. In particular, the test on the one hand indicates that two independent equations should be applied for temporary migration, and on the other it supports the use of a bivariate model for estimating the effect of social capital on permanent migration. In the latter case, the estimated correlation across the errors in the two equations suggests that the unobservables that determine social capital and permanent migration are negatively correlated. Given the significant correlation, a single probit model for permanent migration yields biased estimates as it misleadingly attributes part of the effect of the unobservables to social capital. The effect of the unobservables appears to counteract the social capital effect, producing a downward bias in the coefficient from a single probit model.

In summary, the analysis suggests that families participating in social organizations are more likely to send siblings abroad permanently. In particular, the marginal effect based on the use of the marginal probabilities indicates that participation in social groups augments the probability of permanent migration by 11 percentage points (Table A5). It is possible that families receive from the social network important information that is crucial for emigration and hence social capital performs a role that complements permanent emigration. A different picture emerges for temporary migration. Social capital is associated with a lower probability of moving temporarily. This may indicate that families rely on social capital rather than temporary circular migration to overcome contingent liquidity constraint and therefore social capital is a substitute for temporary migration. The preliminary descriptive analysis in Section 3, reveals that the information and help for temporary migration is provided by family and friends at destinations, rather than by contacts in Albania. This corroborates the hypothesis that social capital does not represent an informative resource for temporary moves.

The theory on social capital suggests that the relationship between social capital and migration may be twofold. On the one hand, as just assessed, social capital influences migration, but on the other migration may influence social capital in that "social capital depreciates when people leave their community" (Glaeser et al. 2002). In this regard, a variable proxing for the community level of migration is introduced. The variable that is designed to proxy for this effect captures the existence of out-migration from the community. This variable is computed from a qualitative question rather than through counting the effective proportion of out-migration from the community. Table 7 reports the estimates.

The community migration variable has the expected sign, as a higher community migration reduces social capital. In particular, the computed marginal effect based on the marginal probability reveals that living in communities which experience out-migration reduces the probability of group participation by four percentage points. This effect is consistent with the evidence in Glaeser et al. 2002, where mobility through migration negatively influences the number of organizations in which one participates. It is worth noting that the estimated coefficients of the other covariates are robust to the inclusion of the community migration variable, as the coefficients remain stable in terms of both its magnitude and significance.

### TABLE 7: HERE

# 7.3 Types of organization

The group participation measure did not distinguish between the types of association a household belongs to. However, it has been recognized that some organizations are characterized by a higher co-operative nature than others. For example, Knack and Keefer (1997) discriminate religious, education, art and youth organizations on the one hand, and welfare, trade union and professional ones on the other. The latter appear to act as rent-seeking organizations, whereas the former are more prone to involve social interactions, which are beneficial for trust building and co-operative behaviour.

This distinction would ideally enhance the previous analysis, in that the link between migration and membership in co-operative groups can be assessed. Unfortunately, the decomposition of social capital into sub-groups introduces a problem of small cell sizes, as the joint event of migration and sub-groups participation occurs in a very limited number of cases. For this reason, the analysis cannot be developed further.

-

<sup>&</sup>lt;sup>10</sup> This variable is drawn from a community questionnaire where the questions are administered to a group of persons reputed to be best informed within a community.

#### **8 Summary and Conclusion**

The paper studies the relationship between social capital and household out-migration from Albania. In particular the analysis attempted to determine whether the participation in social organizations, which are commonly defined as a form of social capital, represents a complement or a substitute for emigration. The nature of the relationship depends on the motivations behind the two choices, which induce the households to join a group and to invest in migration.

In the case where the two decisions are driven by common underlying factors, such as risk reduction or the relaxation of financial constraints, it can be expected that social capital and migration act as substitutes for one another. In this perspective, participation in social organizations and migration are alternative mechanisms to achieve one goal. On the contrary, insofar as the decision to participate in social capital activities reflects the need to share information, migrant households may benefit from the existence of such informative networks and social groups can act as a channel to spread important information that facilitates migration. Under this circumstance, migration and social capital may perform a role of complements for one another.

Two types of international migration have been analysed in this study. These are short period temporary migration of the household head and permanent migration, characterised by a longer duration abroad and eventually no-return to the place of origin. The empirical analysis based on a univariate probit suggests that the participation in social organizations reduces the probability of temporary migration by 0.4 of a percentage point, on average and *ceteris paribus*. This negative relationship can derive from the fact that social capital substitutes for temporary migration. In Albania temporary migration is often characterised by frequent back and forth movements between host and home countries, which may occur in case of liquidity needs. Families in case of adverse shocks eventually recourse to social capital and treat it as an alternative to temporary migration, in order to overcome contingent liquidity constraints.

A second finding is that the error terms in the permanent migration and in the group participation functions are correlated. Therefore, the casual effect of social capital on permanent migration may be driven by unobservable common features which, in the first instance, determine the participation in a specific group and influence the behaviour within the group. On the contrary, such a correlation is not evident for temporary migration. The bivariate probit estimation for permanent migration reveals that families participating in social organizations are more likely to send siblings abroad permanently. The participation in social groups augments the probability of permanent migration by 11 percentage points. It is possible that families receive from the social network important information that is crucial for permanent emigration of the siblings and hence social capital performs a role as complement to permanent emigration.

Other empirical findings are that education has either no or even a negative effect on migration, suggesting that human capital is not a critical asset for temporary and for permanent migration in the Albanian context. Wealthier households are less likely to migrate, both permanently and temporarily, but the magnitude of the effect appears negligible. Well-off families have eventually less need of capital through migration, and only those families living in poverty recourse to migration as a means to escape from it. Controlling for wealth, the ownership of land positively influences the probability of both types of migration. This may indicate that land ownership increases the need for capital, to engage productive activities in rural areas.

In agreement with the theoretical background, participation in social groups is found to reduce in more fragmented communities.

Social capital is a valuable asset for households, as informal arrangements are among the set of strategies that households employ to cope with risks under weak formal institutions. This asset represents a risk free and cost-free mechanism that helps families achieve a better and more secure standard of living. On the contrary, despite its large benefits, migration imposes heavy social costs. The separation of families due to migration may create problems to all generations of children, parents and grandparents. The elderly lose their social support, while women feel great psychological stress, consequent on the emigration of men in their household. Insofar as social capital represents a substitute strategy, at least to temporary migration, its benefits should not be underestimated. Albania faces a crucial problem in this regard, as the disintegration of the old agricultural and industrial networks has limited the availability of social organizations and therefore has constrained the possibility for social capital to properly develop. Alternative economic organizations should be developed to replace the old ones, with a clear statement of their mission. This would help Albanians overcome their sceptical feeling against associations, which remind them the old times of the socialist cooperatives.

#### References

Alesina A.; R. Baqir and W. Easterly (1999) "Public Goods And Ethnic Divisions" *The Quarterly Journal of Economics*, vol. 114(4), pp 1243-1284

Alesina A. and E. La Ferrara (2000) "Participation in heterogeneous communities" *The Quarterly Journal of Economics*, vol. 115(3), pp. 847-904

Barr A. (2000) "Social Capital and Technical Information Flows in the Ghanaian Manufacturing Sector" *Oxford Economic Papers*, vol. 52(3), pp 539-59

Besley T. (1995) "Nonmarket Institutions for Credit and Risk Sharing in Low-Income Countries" *Journal of Economic Perspectives*, vol 9(3), pp 115-27

Beugelsdijk S. and T. van Schaik (2005) "Social capital and growth in European regions: an empirical test" *European Journal of Political Economy*, vol 21(2), pp 301-324

Bramoullé Y. and R. Kranton (2007) "Risk Sharing Across Communities" *American Economic Review*, vol. 97(2), pp 70-74

Carletto C., B. Davis, M. Stampini and A. Zezza (2006) "Internal Mobility and International Migration in Albania", *International Migration Review*, vol. 40 (4), Winter

Coleman J. (1988) "Social Capital in the Creation of Human Capital" *American Journal of Sociology*, vol. 94, pp. 95-120

Costa D. L. and M. E. Kahn (2003) "Understanding the American Decline in Social Capital, 1952-1998" *Kyklos*, vol 56(1), pp 17-46

De Weerdt J. and S. Dercon (2006) "Risk-sharing networks and insurance against illness" *Journal of Development Economics*, vol. 81(2), pages 337-356

Deaton A. (1992) *Understanding Consumption*, Clarendon Press, Oxford

Djajic S. and R. Milbourne (1988) "A General Equilibrium Model of Guest-Worker Migration: A Source-Country Perspective" *Journal of International Economics*, vol. 25, pp.335-351

Durlauf S. N. (2002) "On the Empirics of Social Capital" *The Economic Journal*, vol. 112, pp. F459-479

Dustmann C. (2001) "Why go back? Return Motives of Migrant Workers" in S. Djajic (ed), *International Migration: Trends, Policy, and Economic Impacts.* Routledge

Dustmann C. (1997) "Return migration, uncertainty and precautionary savings" *Journal of Development Economics*, vol. 52, pp. 295-316

Dustmann C. and Y. Weiss (2007) "Return Migration: Theory and Empirical Evidence for the UK" *British Journal of Industrial Economics*, vol. 45 (2), pp. 236-256

Easterly W. and R. Levine (1997) "Africa's Growth Tragedy: Policies and Ethnic Divisions" *The Quarterly Journal of Economics*, vol. 112(4), pp 1203-50.

Ethan L; J. P. Thomas and T. Worrall (2002) "Mutual Insurance, Individual Savings and Limited Commitment" *Review of Economic Studies*, vol. 69, pp. 115-139

Fafchamps M. (1997) "Trade Credit in Zimbabwean Manufacturing" World Development, vol. 25(5), pp 795-815

Fafchamps M. and F. Gubert (2007a) "The formation of risk sharing networks" *Journal of Development Economics*, vol. 83(2), pp 326–350

Fafchamps M. and F. Gubert (2007b) "Risk sharing and network formation" *American Economic Review*, vol. 97(2), pp. 75-79

Fafchamps M. and S. Lund (2003) "Risk Sharing Networks in Rural Philippines" *Journal of Development Economics*, vol 71, pp 261-287

Gertler P. and J. Gruber (2002) "Insuring Consumption Against Illness" *American Economic Review*, vol. 92, pp. 51-70

Glaeser E. L., D. Laibson and B. Sacerdote (2002) "An Economic Approach to Social Capital" *Economic Journal*, vol 112(483), pp 437-458

Glaeser E. L., D. I. Laibson, J. A. Scheinkman and C. L. Soutter (2000) "Measuring Trust" *The Quarterly Journal of Economics*, vol 115(3), pp 811-846

Granovetter M. (1995) Getting a Job: A Study of Contacts and Careers. Chicago: University of Chicago Press

Greene W. H. (2000) Econometric Analysis, Prentice-Hall

Greene W. H. (1998) "Gender Economic Courses in Liberal Arts Colleges: Comment" *Journal of Economic Education*, vol. 29(4), pp. 291-300

Greene, W.H. (1996) "Marginal Effects in the Bivariate Probit Model" *Working Papers 96-11*, New York University, Leonard N. Stern School of Business, Department of Economics

Grootaert C. (2000), "Social Capital, Household Welfare, and Poverty in Indonesia" World Bank Policy Reserach Working Paper n 2148, The World Bank, Washington D.C.

Grootaert C., G. Oh and A. Swamy (2002) "Social Capital, Household Welfare and Poverty in Burkina Faso" *Journal of African Economies*, vol. 11(1), pages 4-38

Halliday T. (2008) "Migration, Risk and the Intra-Household Allocation of Labor in El Salvador" *IZA Discussion Papers 3322*, Institute for the Study of Labor (IZA) Bonn

Halliday T. (2006) "Migration, Risk and Liquidity Constraints in El Salvador" *Economic Development and Cultural Change*, vol. 54(4), pp. 893-925

Hatton T.J. and J.G. Williamson (2004) "International Migration in the Long Run: Positive Selection, Negative Selection and Policy" *IZA Discussion Paper.1304*, Institute for the Study of Labor (IZA) Bonn

Hill J. K. (1987) "Immigrant Decisions Concerning Duration of Stay and Migration Frequency" *Journal of Development Economics*, vol. 25, pp. 221-234

Jalan J. and M. Ravallion (1999) "Are the poor less well-insured? Evidence on vulnerability to income risk in rural China" *Journal of Development Economics*, vol. 58, pp. 61-81

King R. (2005) "Albania as a laboratory for the study of migration and development" *Journal of Southern Europe and the Balkans*, vol 7(2), pp 133-155

Knack S and P. Keefer (1997) "Does Social Capital Have an Economic Payoff? A Cross-Country Investigation" *The Quarterly Journal of Economics*, vol. 112(4), pp 1251-88

La Ferrara E. (2003) "Kin Groups and Reciprocity: A Model of Credit Transactions in Ghana" *American Economic Review*, vol. 93(5), pp. 1730-1751

La Ferrara E. (2002a) "Inequality and Group Participation: theory and evidence from rural Tanzania. *Journal of Public Economics*, vol. 85, pp. 235-273

La Ferrara E. (2002b) "Self-help Groups and Income Generation in the Informal Settlement of Nairobi" *Journal of African Economies*, vol. 11(1), pp. 61-89

Ligon E., J. P. Thomas and T. Worrall (2002) "Informal insurance arrangements with limited commitment: Theory and evidence from village economies" *The Review of Economic Studies*, vol 69(1), pp 209-244

Lucas, R E B (1987) "Emigration to South Africa's Mines" *American Economic Review*, vol. 77(3), pages 313-30

Lucas R.E. and O. Stark O. (1985) "Motivations to Remit: Evidence from Botswana" *Journal of Political Economy*, Vol. 93 (5) pp. 901-918.

Maluccio J., L. Haddad and J. May (2000) "Social Capital and Household Welfare in South Africa" *Journal of Development Studies*, vol 36 (6), pp 54-81

Massey M. and K. E. Espinosa (1997) "What's driving Mexico-US Migration? A Theoretical, Empirical, and Policy Analysis" *American Journal of Sociology*, vol 102(4), pp 939-999

McKenzie D. (2005) "Measuring inequality with asset indicators" *Journal of Population Economics*, vol. 18(2), pp. 229-260

McKenzie D. and H. Rapoport (2007) "Network effects and the dynamics of migration and inequality: Theory and evidence from Mexico" *Journal of Development Economics*, vol. 84(1), p. 1-24

McKenzie D. J. and H. Rapoport (2006) "Can migration reduce education attainment? Evidence from Mexico" World Bank Policy Research Working Paper 3952, The World Bank, Washington D.C

Miall H. (1995) "Albania: Development and Conflict" report prepared for International Alert, 31 July

Mora J. and J. E. Taylor (2005) "Determinants of Migration, Destination and Sector Choice: Disentangling Individual, Household and Community Effects" in Ozden C. and M. Schiff (ed.), *International Migration, Remittances and the Brain Drain*, World Bank and Palgrave Mcmillan

Mouw T. (2006) "Estimating the causal effect of social capital: a review of recent research" *Annual Review of Sociology*, vol. 32, pp. 79-102

Narayan D. and L. Pritchett (1997) "Cents and sociability: household income and social capital in rural Tanzania" Economic Development and Cultural Change, vol 47(4), pp 871-897

Palloni A.; D. S. Massey; M. Ceballos; K. Espinosa and M. Spittel (2001) "Social Capital and International Migration: A Test Using Information on Family Networks" *The American Journal of Sociolog*, vol. 106(5), pp. 1262-1298

Pinger P. (2007) "Come Back or Stay? Spend Here or There? Temporary versus Permanent Migration and Remittance Patterns in the Republic of Moldova" *Kiel Advanced Studies Working Papers 438*, Kiel Institute for the World Economy

Putnam R.; R. Leonardi and R. Y. Nanetti (1993) "Making Democracy Work: Civic Traditions in Modern Italy" Princeton University Press

Reyes B.I. (2001) "Immigrant Trip Duration: the case of Immigrants from Western Mexico" *International Migration Review*, vol. 35(4), pp.1185-1204

Reyes B.I and L. Mameesh (2002) "Why Does Immigration Trip Duration Vary Across Us Destinations?" *Social Science Quarterly*, vol. 83(2), pp.580-593

Rosenzweig M. R. and O. Stark (1989) "Consumption Smoothing, Migration, and Marriage: Evidence from Rural India" *Journal of Political Economy*, vol 97(4), pp 905-26

Rozelle S.; J. E. Taylor and A. deBrauw (1999) "Migration, Remittances, and Agricultural Productivity in China" *American Economic Review*, vol. 89(2), pp 287-291

Sabates-Wheeler R. and M. Waite (2003) "Albania country brief: property rights and land markets" MIMEO, University of Wisconsin - Madison

Sjaastad L.A. (1962) "The Costs and Returns of Human Migration" *The Journal of Political Economy*, vol 70(5), pp 80-93

Stark O. and D. Bloom (1985) "The New Economics of Labor Migration" *American Economic Review*, vol. 75(2), pp. 173-78

Stark O., C. Helmenstein and Y. Yegorov (1997) "Migrants' Savings, Purchasing Power Parity, and the Optimal Duration of Migration" *International Tax and Public Finance*, vol. 4, pp. 307-324

Stark. O. and R.E.B Lucas (1988) "Migration, Remittances, and the Family" *Economic Development and Cultural Change*", vol. 36(3), pp 465-81

Stark O. and J. E. Taylor (1991)"Migration Incentives, Migration Types: The Role of Relative Deprivation" *Economic Journal*, vol. 101(408), pp 1163-78

Stecklov G., C. Carletto, C. Azzarri and B. Davis (2008) "Agency, Education and Networks: Gender and International Migration from Albania" World *Bank Policy Research Working Paper n.* 4507, The World Bank, Washington D.C.

Taylor J. E. (1992) "Remittances and inequality reconsidered: Direct, indirect, and intertemporal effects" *Journal of Policy Modeling*, vol. 14(2), pages 187-208

Taylor J. E. (1987) "Undocumented Mexico-U.S: Migration and the Returns to Households in Rural Mexico" *American Journal of Agricultural Economics*, vol 69(3), pp 626-638

Taylor, J. and Martin P. (2001) "Human capital: Migration and rural population change" in: B. L. Gardner and G. C. Rausser (ed.), *Handbook of Agricultural Economics*, Elsevier

Taylor J. E; S. Rozelle and A. de Brauw (2003) "Migration and Incomes in Source Communities: A New Economics of Migration Perspective from China" *Economic Development and Cultural Change*, vol. 52(1), pp 75-101

Taylor J. E. and T.J. Wyatt (1996) "The shadow value of migrant remittances, income and inequality in a household-farm economy" *Journal of Development Studies*, vol. 32(6), pp. 899-912

Tiepoh, M.; N. Geepu and B. Reimer (2004) "Social capital, information flows, and income creation in rural Canada: a cross-community analysis" *The Journal of Socio-Economics*, vol. 33(4), pp 427-448

Townsend R. (1994) "Risk and Insurance in Village India" Econometrica, vol. 62, pp. 539-591

Vullnetari J. (2007) "Albanian migration and development: state of the art review" *IMISCOE Working Paper n. 18* 

Winters P., A. de Janvry and E. Sadoulet (2001) "Family and Community Networks in Mexico-U.S. Migration" *The Journal of Human Resources*, vol 36(1), pp 159-184

World Bank (2007) "Albania - Urban growth, migration and poverty reduction: a poverty assessment" *Report No. 40071-AL*, The World Bank, Washington DC

World Bank (2002) "Poverty in Albania: a Qualitative Assessment" World Bank Technical Paper n. 520, The World Bank, Washington D.C.

Yang D. and H. Choi (2007) "Are Remittances Insurance? Evidence from Rainfall shocks in the Philippines" *World Bank Economic Review*, vol. 21(2), pp. 219-248

Yang D. (2006) "Why Do Migrants Return to Poor Countries? Evidence from Philippine Migrants' Responses to Exchange Rate Shocks" *The Review of Economics and Statistics*, vol. 88(4), pp 715-735

# Appendix A

Table A1 Summary Statistics and Tests for Differences in Proportions. Temporary Migration

	Temporary Migrants $(p_M)$	Non-migrants $(p_N)$	z-scores
Male	0.99	0.91	3.48***
Education attainment			
Primary	0.59	0.45	3.43***
Secondary	0.15	0.18	-0.96
Vocational	0.20	0.25	-1.41
University and post	0.06	0.12	-2.28***
Location			
Coastal	0.20	0.26	-1.67*
Central	0.32	0.24	2.27***
Tirana	0.12	0.25	-3.70***
Mountain	0.36	0.25	3.07***
Urban	0.40	0.60	-4.96***

*Notes:*\*\*\* denotes statistical significance at 1% level. \*\*denotes statistical significance at 5% level. \* denotes statistical significance at 10% level using two tailed tests. The proportions are calculated for migrants and non-migrants separately. To receive 100% one should add vertically the proportions within each group of variables.

The non-parametric t-test is computed as:  $[p_M - p_{NM}] / [\overline{p} (1-\overline{p})/n_M + \overline{p} (1-\overline{p})/n_{NM}]^{1/2} \sim N(0, 1)$ 

where  $p_M$  and  $p_{NM}$  represents, respectively, the proportion of migrant and non-migrant families in each category;  $\overline{p}$  represents the fraction of families in each category: it is computed as the sum of the absolute number of migrant families and the absolute number of non-migrant families for each category divided by the total number of families.  $n_M$  denotes the number of migrant families while  $n_{NM}$  denotes the number of non-migrant families, which is respectively 157 and 2,816.

Table A2: Summary Statistics and Tests for Differences in Proportions. Permanent Migration

	Permanent Migrants $(p_M)$	Non-migrants $(p_N)$	z-scores
Male	0.89	0.91	-0.75
Education attainment			
Primary	0.72	0.45	5.86***
Secondary	0.02	0.18	-4.59***
Vocational	0.16	0.25	-2.27***
University and post	0.11	0.12	-0.33
Location			
Coastal	0.4	0.26	3.44***
Central	0.17	0.24	-1.79**
Tirana	0.19	0.25	-1.51
Mountain	0.24	0.25	-0.25
Urban	0.43	0.60	-3.73***

Notes: see notes to Table A1. The number of migrant families and non-migrant families which is respectively 122 and 2816

Table A3: Summary statistics of the variables entering the non-productive wealth index

		Standard		
	Mean	Deviation	Min	Max
Area of dwelling	2.79	0.99	1	5
Water Facilities	0.62	0.49	0	1
Sanitary Facilities	0.74	0.44	0	1
Phone	0.29	0.46	0	1
Coloured Television	1.07	0.45	0	4
Terrace	0.67	0.47	0	1

Notes: the dwelling area variable is characterised as follows. 1 stands for less than 40 sq metres area; 2 for 40-69 sq metres; 3 for 70- sq metres; 4 for 100- sq metres; 5 for more than 130 sq metres.

Table A4: Summary statistics and description of the variables

Variable	Description	Mean	Std. Dev.
Temporary Migration	=1 if household head migrated abroad temporarily after 2004; 0 otherwise	0.05	0.22
Temporary Migration	=1 if the household has a member who migrated abroad temporarily after 2004; 0 otherwise	0.11	0.31
Permanent Migration	=1 if the household has a member who migrated abroad permanently after 2004; 0 otherwise	0.04	0.19
Social Capital	=1 if the household is member of groups, organizations, networks and associations; 0 otherwise	0.22	0.41
Age	Age of Household Head	47.44	9.58
Male	=1 if household head is male; 0 otherwise	0.92	0.28
Education	Number of years of completed schooling of household head	10.53	3.75
Coastal	=1 if the household resides in the Central area; 0 otherwise	0.26	0.44
Tirana	=1 if the household resides in Tirana; 0 otherwise	0.24	0.43
Central	=1 if the household resides in the Central area; 0 otherwise	0.24	0.43
Urban	=1 if the family resides in an urban settlement; 0 otherwise	0.58	0.49
Land	Total area of agriculture plot. Thousands square meters	4.52	19.77
Land Squared	Square of land	411.24	9845.63
Asset Index	Index of assets owned in 1990. Principal component method	0.01	1.45
Instruments			
Activity			
Fragmentation	Index of Activity Fragmentation	0.80	0.06
Ethnicity Fragmentation	Index of Ethnic Fragmentation	0.03	0.06
Community Migration	=1 if the community experienced occasional or a lot of migration abroad; 0 otherwise	0.80	0.40

Table A5: Marginal and Impact Effects in the migration equations

	Pre	Probit		SURE				
	Temporary	Permanent	Temporary	Permanent	Temporary	Permanent	Temporary	Permanent
	Migration	Migration	Migration	Migration	Migration	Migration	Migration	Migration
	Pr (y = 1)	Pr (y = 1)	$Pr(y_1 = 1)$	$Pr(y_1 = 1)$	$Pr(y_1=1; y_2=1)$	$Pr(y_1=1; y_2=1)$	$Pr(y_1   y_2 = 1)$	$Pr(y_1   y_2 = 1)$
Social capital	-0.004	-0.004	0.054	0.099	0.001	0.003	0.004	0.016
Age	-0.001	0.002	-0.002	0.002	-0.00001	0.0001	-0.00004	0.0003
Male	0.010	-0.013	0.015	-0.018	0.0001	-0.0001	0.0004	-0.001
Education	-0.0004	-0.001	-0.001	-0.002	0.00001	0.00004	0.000004	0.00003
Coastal	-0.004	0.008	-0.007	0.006	-0.00001	0.0005	-0.0001	0.001
Tirana	-0.007	0.003	-0.010	0.005	-0.0001	0.00004	-0.0003	0.0003
Central	0.002	-0.009	0.001	-0.015	0.0001	0.0001	0.0003	-0.0005
Urban	0.0003	0.004	-0.001	0.002	0.00003	0.0002	0.0001	0.001
Land	0.001	0.001	0.001	0.001	0.0001	0.0002	0.001	-0.002
Asset Index	-0.002	-0.006	-0.002	-0.006	-0.00002	-0.0002	-0.0001	-0.001
Activity Fragmentation	-	-	-	-	-0.0001	-0.001	-0.0004	-0.002
<b>Ethnicity Fragmentation</b>	-	-	-	-	-0.0002	-0.001	-0.001	-0.003

Notes: columns (1) and (2) produces the marginal effect of single equation migration probit. Columns (3) and (4) produce the marginal effect based on the marginal probability in a bivariate probit model, where  $y_1$  refers to migration outcome. Columns (5) and (6) produce the marginal effect on the joint probability of migration and group participation, where  $y_1$  refers to migration outcome and  $y_2$  to group participaton. Columns (7) and (8) produce the marginal effect on the conditional probability

The marginal effect of land in the probit is computed as:

 $\partial$  Prob (y=1) /  $\partial$  L=  $\phi$  (X' $\beta$ ) \* ( $\beta_L$  + 2  $\beta_{Lsq}$  \*  $\overline{L}$  ), where  $\beta_L$  is the coefficient of land,  $\beta_{Lsq}$  is the coefficient of land squared,  $\overline{L}$  is the sample mean of land,  $\phi$  (.) is the pdf and X' $\beta$  is the linear prediction.

The marginal effect of land based on the marginal in the bivariate probit is computed as:

 $\partial$  Prob (y=1) /  $\partial$  L=  $\phi$  (X<sub>1</sub>' $\beta$ <sub>1</sub>) \* ( $\beta$ <sub>1L</sub> + 2  $\beta$ <sub>1Lsq</sub> \*  $\overline{L}$ ), where X<sub>1</sub>' $\beta$ <sub>1</sub>,  $\beta$ <sub>1L</sub> and  $\beta$ <sub>2L</sub> are respectively the linear prediction, the coefficient of land, and the coefficient of land squared in the migration equation.

The marginal effect of land on the joint probability in the biprobit model is given by:

 $\partial$  Prob  $(y_1=1, y_2=1) / \partial L= \{.\}$   $(\beta_{1L}+2 \beta_{1Lsq}*\overline{L})+\{.\}$   $(\beta_{2L}+2 \beta_{2Lsq}*\overline{L})$ , where the curly brackets correspond to the curly brackets in equation (2) of Appendix B, where  $\beta_{1L}$  and  $\beta_{2L}$  are the coefficient of land respectively in the migration and social capital equations,  $\beta_{1Lsq}$  and  $\beta_{2Lsq}$  are the coefficient of land squared in the two equations. The marginal effect of land on the conditional probability in the biprobit model is given by:

 $\partial$  Prob  $(y_1=1|y_2=1)/\partial L=\{.\}$   $(\beta_{1L}+2\beta_{1Lsq}*\overline{L})+\{.\}$   $(\beta_{2L}+2\beta_{2Lsq}*\overline{L})$ , where the curly brackets correspond to the curly brackets in equation (3) of Appendix B

# Appendix B

# **Computation of the marginal Effects**

The bivariate binary choice model is given by:

$$y_1^* = \mathbf{x}_1' \mathbf{\beta}_1 + \mathbf{u}_1$$

$$y_2^* = \mathbf{x}_2' \mathbf{\beta}_2 + \mathbf{u}_2$$

where  $y_1$  and  $y_2$  are dichotomous variables observed according to the rule:

$$y_j = 1 \text{ if } y_j^* > 0 \text{ and } y_j = 0 \text{ if } y_j^* \le 0, j=1, 2$$

and  $u_1$  and  $u_2$  are bivariate normal, with zero means, unit variance and covariance  $\rho$ .

There are several marginal effects can be computed in a bivariate probit model (Greene, 1996; Greene, 1998)

i) Given the marginal probability of  $y_1=1$ :

$$Prob(y_1=1| \mathbf{X}_1, \mathbf{X}_2) = \Phi(z_1)$$

the marginal effect on the marginal probability is given by:

$$\frac{\partial \Phi}{\partial x} = \phi(z_i)\beta_{1x} \tag{1}$$

ii) Given the bivariate probability of positive outcomes, namely  $y_1=1$  and  $y_2=1$ :

$$Prob(y_1=1, y_2=1 | \mathbf{x_1}, \mathbf{x_2}) = \Phi_{BVN}(z_1, z_2, \rho)$$

where 
$$z_1 = \mathbf{x}_1' \mathbf{\beta}_1$$
 and  $z_2 = \mathbf{x}_2' \mathbf{\beta}_2$ 

the marginal effect on the joint probability is given by:

$$\frac{\partial \Phi_{BVN}}{\partial x} = \{ \phi(z_1) * \Phi[(z_2 - \rho(z_1)) / \sqrt{1 - \rho^2} ] \} \beta_{1x} + \{ \phi(z_2) * \Phi[(z_1 - \rho(z_2) / \sqrt{1 - \rho^2} ] \} \beta_{2x}$$
 (2)

iii) Given the probability of  $y_1=1$  conditional on  $y_2=1$ 

Prob 
$$(y_1=1|\ y_2=1,\ \mathbf{x_1},\ \mathbf{x_2})$$
 = Prob  $(y_1=1,\ y_2=1|\ \mathbf{x_1},\ \mathbf{x_2})$  / Prob $(y_2=1|\ \mathbf{x_1},\ \mathbf{x_2})$   
=  $\Phi_{BVN}(z_1,\ z_2,\ \rho)$  /  $\Phi(z_2)$ 

the marginal effect on the conditional probability is given by:

$$\frac{\partial \Pr{ob(y_1 \mid y_2 = 1, \mathbf{X_1, X_2})}}{\partial x} = \left\{ \frac{g_1}{\Phi(z_2)} \right\} \beta_{1x} + \left\{ \frac{1}{\Phi(z_2)} (g_2 - \Phi_{BVN} \frac{\phi(z_2)}{\Phi(z_2)}) \right\} \beta_{2x}$$
(3)

where  $g_1 = \{\phi(z_1)\Phi[(z_2 - \rho(z_1)/\sqrt{1-\rho^2}]\}$  and the subscripts 1 and 2 are reversed to obtain  $g_2$ 

Table 1: Household characteristics of recent temporary/permanent international migrants. 2004

		Temporary	Permanent	No Migration
	Total	0.05	0.04	0.91
	Gender of head:			
	Male	0.99	0.89	0.91
	Education attainment of the head:			
	Primary	0.59	0.72	0.45
	Secondary	0.15	0.02	0.18
	Vocational	0.20	0.16	0.25
% of households	University and post	0.06	0.11	0.12
in temporary/permanent/	Location:			
no migration condition	Coastal	0.2	0.4	0.26
	Central	0.32	0.17	0.24
	Tirana	0.12	0.19	0.25
	Mountain	0.36	0.24	0.25
	Urban	0.40	0.43	0.60
Summary	Education Years of head	10	9	11
Total		157	122	2816

Notes: the proportions are calculated for temporary, permanent and non-migrants separately. To receive 100% one should add vertically the proportions within each group of variables. The number of families in the temporary, permanent and no migration categories are respectively 157, 122 and 2,816. The association membership pertains to the following type of groups: labour related, village, cultural, religious, environmental, youth, veterans, sport, ethnic or other.

Table 2: Selected characteristics of recent temporary international migration. 2004

Variables	Desciption	Percent
Duration of the stay:	less then 6 months	68
Reason for migration	To work, look for work	92
Prearranged job	Yes	70
Stayed abroad with spouse	Yes	4
Stayed abroad with children	Yes	6
Reason for returning Albania	Could not get residence	5
	Could not get work	4
	Permit expired	9
	No intention to stay	6
	Accumulated enough money	2
	Seasonal work	31
	Got expelled	4
	Family reasons	17
	Other	21
Planning to migrate again	yes	82
Migrated before	Yes	90
Duration previous migration	less then 6 months	63
Provider of information	Family, relatives or friends in Albania	11
	Family, relatives or friends abroad	34
	Previous personal experience	52
	Other	3
Total		157

**Table 3: Most important associations** 

Type of association	Percent	Number of households
Labour Associations	36.09	244
Political Associations	15.68	106
Neighbourhood /village council of dignitaries	10.8	73
Religious groups	7.69	52
Youth groups	4.44	30
Cultural association	2.81	19
Sports group	2.07	14
Association for water supply	1.63	11
Veterans associations	1.63	11
Ethnic-based community group	1.63	11
Association for environment protection	1.04	7
NGO	0.74	5
Association for the consumers' protection	0.3	2
Other	13.46	91
Total		676

**Table 4: Social Capital and Migration** 

•	Tempora	Temporary		nent	No-migration	
	Percent	Number	Percent	Number	Percent	Number
Full Sample	0.05	157	0.04	122	0.91	2816
Association membership	0.03	21	0.04	25	0.93	630
No association membership	0.06	136	0.04	97	0.90	2185
z-score	-2.64		-0.37		2.27	

Notes: in the sample 676 households participate in social groups, while 2,418 do not participate. The non-parametric z-score is performed on the difference in proportion between the association membership and the non-membership, within the different migration categories. See the note to Table A1 for the statistics of the test.

Table 5: Household head temporary migration. Probit and SURE estimations

**Probit Temporary** Social Temporary Social Temporary Social Temporary Variables Migration Capital Migration Capital Migration Capital Migration Social -0.215\* 1.003 0.959 0.879 Capital (0.112)(0.891)(0.778)(0.841)-0.045\*\*\* -0.046\*\*\* -0.046\*\*\* 0.014\*\*\*-0.046\*\*\* Age 0.014\*\*\* 0.014\*\*\* (0.005)(0.005)(0.003)(0.005)(0.003)(0.005)(0.003)Male 0.986\*\*\* 0.207\*\* 0.931\*\*\* 0.213\*\* 0.931\*\*\* 0.212\*\* 0.935\*\*\* (0.367)(0.104)(0.361)(0.104)(0.361)(0.104)(0.362)Education -0.016 0.070\*\*\* -0.031\*\* 0.071\*\*\* -0.031\*\* 0.070\*\*\* -0.030\*\* (0.013)(0.008)(0.015)(800.0)(0.015)(0.008)(0.015)Coastal 0.233\*\*\* 0.279\*\*\* -0.223\* -0.173-0.225\* 0.243\*\*\* -0.221\* (0.120)(0.076)(0.120)(0.075)(0.119)(0.077)(0.121)-0.375\*\*\* Tirana -0.041 -0.341\*\* -0.016 -0.342\*\* -0.044 -0.345\*\* (0.140)(0.142)(0.141)(0.141)(0.084)(0.082)(0.084)Central 0.086 0.328\*\*\* 0.013 0.398\*\*\* 0.014 0.372\*\*\* 0.017 (0.105)(0.079)(0.099)(0.077)(0.098)(0.079)(0.100)Urban 0.014 0.214\*\*\* -0.026 0.208\*\*\* -0.025 0.209\*\*\* -0.025 (0.125)(0.071)(0.116)(0.071)(0.117)(0.071)(0.118)Land 0.045\*\*0.011\*\*\* 0.040\*\* 0.011\*\*\* 0.041\*\*0.011\*\*\*0.041\*(0.020)(0.004)(0.019)(0.004)(0.019)(0.004)(0.019)Land Squared -0.00002\*\* -0.00002\*\* -0.00002\*\* -0.001\* -0.001\* -0.001\* -0.001\* (0.001)(0.00001)(0.001)(0.00001)(0.001)(0.00001)(0.001)-0.086\*\* Asset -0.048\*\* -0.072\* -0.047\*\* -0.072\* -0.041\* -0.073\* Index (0.044)(0.023)(0.042)(0.023)(0.042)(0.023)(0.042)Activity -0.646 -0.713\* Fragmentation (0.417)(0.419)Ethnicity -1.046\*\* -1.084\*\* Fragmentation (0.428)(0.426)Constant -0.435 -2.210\*\*\* -0.248-2.742\*\*\* -0.250 -2.126\*\*\* -0.256 (0.209)(0.481)(0.424)(0.468)(0.468)(0.425)(0.472)N 3094 3094 3094 3094 3094 3094 3094 Pseudo R2 0.14 Rho -0.704 -0.676 -0.625(0.574)(0.492)(0.517)Wald test of rho=0 Chi2(1)=1.50Chi2(1)=1.89 Chi2(1)=1.46P-value: 0.22 P-value: 0.17 P-value: 0.23 Orthogonality of Chi2(1)=0.06Chi2(1)=0.95Chi2(2)=1.03instrument P-value: 0.80 P-value: 0.33 P-value: 0.60

Notes: robust standard error in parenthesis. \* denotes significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

Table 6: Permanent migration. Probit and SURE estimations

**Probit SURE** Permanent Social Permanent Social Permanent Social Permanent Variables Migration Capital Migration Capital Migration Capital Migration 0.789\* 0.796\* 0.888\* Social -0.077 Capital (0.473)(0.453)(0.458)(0.111)0.036\*\*\* 0.036\*\*\* 0.036\*\*\* Age 0.041\*\* 0.014\*\*\* 0.014\*\*\* 0.014\*\* (0.004)(0.003)(0.005)(0.003)(0.005)(0.003)(0.005)Male -0.195 0.208\*\* -0.219 0.215\*\*-0.220 0.213\*\* -0.221 (0.104)(0.151)(0.148)(0.104)(0.148)(0.104)(0.147)Education 0.071\*\*\* -0.025\* 0.072\*\*\* -0.025\* 0.071\*\*\* -0.026\*\* -0.012 (0.013)(0.008)(0.013)(0.013)(0.013)(0.008)(800.0)Coastal 0.137 0.237\*\*\* 0.087 0.289\*\*\* 0.0880.247\*\*\* 0.083 (0.124)(0.076)(0.123)(0.075)(0.123)(0.076)(0.122)Tirana 0.056 -0.047 0.068 -0.018 0.069 -0.047 0.071 (0.139)(0.082)(0.134)(0.081)(0.134)(0.082)(0.133)Central 0.318\*\*\* -0.235\* 0.395\*\*\* -0.237\* 0.367\*\*\* -0.241\* -0.185(0.135)(0.076)(0.130)(0.076)(0.130)(0.077)(0.129)Urban 0.081 0.203\*\*\* 0.036 0.197\*\*\*0.0360.199\*\*\* 0.031 (0.118)(0.069)(0.114)(0.069)(0.114)(0.069)(0.113)Land 0.026\*\*\* 0.010\*\*\* 0.022\*\*\* 0.011\*\*0.023\*\* 0.010\*0.022\*\*(800.0)(0.004)(0.009)(0.004)(0.009)(0.004)(0.009)-0.0002 Land Squared -0.00002\* -0.0002 -0.00002\* -0.0002 -0.00002\* -0.0002(0.0001)(0.00001(0.0001)(0.00001(0.0001)(0.00001)(0.0001)Asset -0.106\*\* -0.044\* -0.092\*\* -0.044\*\* -0.092\*\* -0.038 -0.091\*\* Index (0.042)(0.023)(0.041)(0.022)(0.041)(0.023)(0.041)Activity -0.824\*\* -0.757\* Fragmentation (0.421)(0.418)Ethnicity -1.105\*\* -1.147\*\*\* Fragmentation (0.430)(0.435)Constant -3.706\*\*\* -2.109\*\*\* -3.365\*\*\* -2.735\*\*\* -3.360\*\*\* -2.023\*\*\* -3.320\*\*\* (0.347)(0.422)(0.366)(0.209)(0.365)(0.421)(0.362)N 3094 3094 3094 3094 3094 3094 3094 Pseudo R2 0.11 -0.502\* Rho -0.507\* -0.566\*\* (0.276)(0.264)(0.275)Wald test of rho=0 Chi2(1) = 3.30Chi2(1) = 3.68Chi2(1) = 4.22P-value: 0.07 P-value: 0.06 P-value: 0.04 Orthogonality of Chi2(1) = 1.00Chi2(1) = 1.10Chi2(2) = 1.90instrument P-value: 0.32 P-value: 0.29 P-value: 0.39

Notes: robust standard error in parenthesis. \* denotes significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

 $\begin{tabular}{ll} \textbf{Table 7: SURE estimations with control for community out-migration. Temporary and permanent migration} \\ \textbf{SURE} \end{tabular}$ 

	G 1	T	G 1	301		D	G 1	D
Variables	Social	Temporary	Social	Temporary	Social	Permanent		Permanent
G : 1	Capital	Migration	Capital	Migration	Capital	Migration	Capital	Migration
Social		1.124		1.028		0.775*		0.874**
Capital	0.04 % division	(0.829)	0.04.4.5.5.5	(0.842)	0.04.435355	(0.449)	0.04.4.5.5.5	(0.433)
Age	0.015***	-0.046***	0.014***	-0.046***	0.014***	0.036***	0.014***	0.036***
	(0.003)	(0.005)	(0.003)	(0.005)	(0.003)	(0.005)	(0.003)	(0.005)
Male	0.205**	0.928***	0.210**	0.931***	0.206**	-0.218	0.210**	-0.220
	(0.104)	(0.360)	(0.104)	(0.361)	(0.104)	(0.148)	(0.104)	(0.147)
Education	0.071***	-0.032**	0.071***	-0.032**	0.072***	-0.025*	0.072***	-0.026**
	(0.008)	(0.015)	(0.008)	(0.015)	(0.008)	(0.013)	(0.008)	(0.013)
Coastal	0.226***	-0.230*	0.236***	-0.226*	0.231***	0.088	0.240***	0.083
	(0.077)	(0.120)	(0.077)	(0.121)	(0.076)	(0.123)	(0.076)	(0.122)
Tirana	-0.110	-0.338**	-0.132	-0.341**	-0.116	0.068	-0.135	0.071
	(0.098)	(0.143)	(0.010)	(0.142)	(0.098)	(0.134)	(0.099)	(0.134)
Central	0.313***	0.010	0.358***	0.013	0.303***	-0.234*	0.352***	-0.240*
	(0.078)	(0.098)	(0.079)	(0.098)	(0.076)	(0.131)	(0.077)	(0.129)
Urban	0.210***	-0.024	0.203***	-0.024	0.197***	0.037	0.191***	0.032
	(0.071)	(0.115)	(0.071)	(0.116)	(0.069)	(0.114)	(0.069)	(0.113)
Land	0.011***	0.040**	0.011***	0.040**	0.010***	0.023***	0.010***	0.022**
	(0.004)	(0.019)	(0.004)	(0.019)	(0.004)	(0.009)	(0.004)	(0.009)
Land	-0.00002**	-0.001*	-0.00002**	-0.001*	-0.00002**	-0.0002	-0.00002**	-0.0002
Squared	(0.00001)	(0.001)	(0.00001)	(0.0007)	(0.00001)	(0.0001)	(0.00001)	(0.0001)
Asset	-0.045*	-0.071*	-0.037	-0.072*	-0.041*	-0.093**	-0.034	-0.091**
Index	(0.023)	(0.041)	(0.023)	(0.042)	(0.023)	(0.041)	(0.023)	(0.041)
Activity	-0.677		-0.755*		-0.793*		-0.872**	
Fragment.	(0.412)		(0.415)		(0.421)		(0.419)	
Ethnicity			-1.172***				-1.240***	
Fragmentat.			(0.428)				(0.435)	
Community	-0.114		-0.144*		-0.112		-0.142*	
Migration	(0.083)		(0.083)		(0.083)		(0.0834)	
Constant	-2.084***	-0.242	-1.963***	-0.246	-1.98***	-3.372***	-1.854***	-3.328***
	(0.426)	(0.463)	(0.429)	(0.466)	(0.431)	(0.364)	(0.430)	(0.360)
N	3094	3094	3094	3094	3094	3094	3094	3094
Rho	-0.788		-0.723		-0.494*		-0.557**	
	(0.565)		(0.549)		(0.260)		(0.258)	
Wald Test	Chi2(1)	= 1.94	Chi2(1)	= 1.73	Chi2(1)	= 3.60	Chi2(1)	= 4.68
of rho=0	i ` ´	ie: 0.16	1	e: 0.19	P-value		P-value	
Orthogonality	<b>:</b>	)=0.06	•	= 1.03	Chi2(1)		Chi2(2)	
of instrument	8	ie: 0.80	· `	e: 0.60	P-value		P-value	

Notes: robust standard error in parenthesis. \* denotes significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%

# NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

#### Fondazione Eni Enrico Mattei Working Paper Series

# Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm http://www.ssrn.com/link/feem.html http://www.repec.org http://agecon.lib.umn.edu http://www.bepress.com/feem/

# **NOTE DI LAVORO PUBLISHED IN 2009**

		NOTE DI LAVORO PUBLISHED IN 2009
SD	1.2009	Michael Hoel: Bush Meets Hotelling: Effects of Improved Renewable Energy Technology on Greenhouse Gas
		<u>Emissions</u>
SD	2.2009	Abay Mulatu, Reyer Gerlagh, Dan Rigby and Ada Wossink: Environmental Regulation and Industry Location
SD	3.2009	Anna Alberini, Stefania Tonin and Margherita Turvani: Rates of Time Preferences for Saving Lives in the
		Hazardous Waste Site Context
SD	4.2009	Elena Ojea, Paulo A.L.D. Nunes and Maria Loureiro: Mapping of Forest Biodiversity Values: A Plural
		<u>Perspective</u>
SD	5.2009	Xavier Pautrel : Macroeconomic Implications of Demography for the Environment: A Life-Cycle Perspective
IM	6.2009	Andrew Ellul, Marco Pagano and Fausto Panunzi: Inheritance Law and Investment in Family Firms
IM	7.2009	Luigi Zingales: The Future of Securities Regulation
SD	8.2009	Carlo Carraro, Emanuele Massetti and Lea Nicita: How Does Climate Policy Affect Technical Change? An
CD	0.2000	Analysis of the Direction and Pace of Technical Progress in a Climate-Economy Model
SD	9.2009	William K. Jaeger: The Welfare Effects of Environmental Taxation
SD	10.2009	Aude Pommeret and Fabien Prieur: <u>Double Irreversibility and Environmental Policy Design</u>
SD	11.2009	Massimiliano Mazzanti and Anna Montini: <u>Regional and Sector Environmental Efficiency Empirical Evidence</u> from Structural Shift-share Analysis of NAMEA data
SD	12.2009	A. Chiabai, C. M. Travisi, H. Ding, A. Markandya and P.A.L.D Nunes: Economic Valuation of Forest
30	12.2005	Ecosystem Services: Methodology and Monetary Estimates
SD	13.2009	Andrea Bigano, Mariaester Cassinelli, Fabio Sferra, Lisa Guarrera, Sohbet Karbuz, Manfred Hafner, Anil
		Markandya and Ståle Navrud: The External Cost of European Crude Oil Imports
SD	14.2009	Valentina Bosetti, Carlo Carraro, Romain Duval, Alessandra Sgobbi and Massimo Tavoni: The Role of R&D
		and Technology Diffusion in Climate Change Mitigation: New Perspectives Using the Witch Model
IM	15.2009	Andrea Beltratti, Marianna Caccavaio and Bernardo Bortolotti: Stock Prices in a Speculative Market: The
		Chinese Split-Share Reform
GC	16.2009	Angelo Antoci, Fabio Sabatini and Mauro Sodini: <u>The Fragility of Social Capital</u>
SD	17.2009	Alexander Golub, Sabine Fuss, Jana Szolgayova and Michael Obersteiner: Effects of Low-cost Offsets on
		Energy Investment - New Perspectives on REDD -
SD	18.2009	Enrica De Cian: Factor-Augmenting Technical Change: An Empirical Assessment
SD	19.2009	Irene Valsecchi: Non-Uniqueness of Equilibria in One-Shot Games of Strategic Communication
SD	20.2009	Dimitra Vouvaki and Anastasios Xeapapadeas: <u>Total Factor Productivity Growth when Factors of Production</u>
CD	21 2000	Generate Environmental Externalities  Civilia Massara Maria Laureira Baula A L.D. Nurses and Bishard Tal. Assarains the Improve of Bishirania.
SD	21.2009	Giulia Macagno, Maria Loureiro, Paulo A.L.D. Nunes and Richard Tol: Assessing the Impact of Biodiversity
IM	22.2009	on Tourism Flows: A model for Tourist Behaviour and its Policy Implications  Bernardo Bortolotti, Veljko Fotak, William Megginson and William Miracky: Sovereign Wealth Fund
IIVI	22.2009	Investment Patterns and Performance
IM	23.2009	Cesare Dosi and Michele Moretto: <u>Auctioning Monopoly Franchises: Award Criteria and Service Launch</u>
	20.2009	Requirements
SD	24.2009	Andrea Bastianin: Modelling Asymmetric Dependence Using Copula Functions: An application to Value-at-
		Risk in the Energy Sector
IM	25.2009	Shai Bernstein, Josh Lerner and Antoinette Schoar: The Investment Strategies of Sovereign Wealth Funds
SD	26.2009	Marc Germain, Henry Tulkens and Alphonse Magnus: Dynamic Core-Theoretic Cooperation in a Two-
		Dimensional International Environmental Model
IM	27.2009	Frank Partnoy: Overdependence on Credit Ratings Was a Primary Cause of the Crisis
SD	28.2009	Frank H. Page Jr and Myrna H. Wooders (lxxxv): Endogenous Network Dynamics
SD	29.2009	Caterina Calsamiglia, Guillaume Haeringer and Flip Klijnb (lxxxv): Constrained School Choice: An
		Experimental Study
SD	30.2009	Gilles Grandjean, Ana Mauleon and Vincent Vannetelbosch (lxxxv): Connections Among Farsighted Agents
SD	31.2009	Antonio Nicoló and Carmelo Rodríguez Álvarez (lxxxv): Feasibility Constraints and Protective Behavior in
C.D.	22.222	Efficient Kidney Exchange
SD	32.2009	Rahmi İlkiliç (lxxxv): Cournot Competition on a Network of Markets and Firms
SD	33.2009	Luca Dall'Asta, Paolo Pin and Abolfazl Ramezanpour (lxxxv): Optimal Equilibria of the Best Shot Game
SD	34.2009	Edoardo Gallo (lxxxv): Small World Networks with Segregation Patterns and Brokers
SD	35.2009	Benjamin Golub and Matthew O. Jackson (lxxxv): How Homophily Affects Learning and Diffusion in
		<u>Networks</u>

SD	36.2009	Markus Kinateder (lxxxv): Team Formation in a Network
SD	37.2009	Constanza Fosco and Friederike Mengel (lxxxv): <u>Cooperation through Imitation and Exclusion in Networks</u>
SD	38.2009	Berno Buechel and Tim Hellmann (lxxxv): <u>Under-connected and Over-connected Networks</u>
SD	39.2009	Alexey Kushnir (lxxxv): Matching Markets with Signals
SD	40.2009	Alessandro Tavoni (lxxxv): Incorporating Fairness Motives into the Impulse Balance Equilibrium and Quantal
30	40.2009	Response Equilibrium Concepts: An Application to 2x2 Games
SD	41.2009	Steven J. Brams and D. Marc Kilgour (lxxxv): <u>Kingmakers and Leaders in Coalition Formation</u>
SD	42.2009	Dotan Persitz (lxxxv): Power in the Heterogeneous Connections Model: The Emergence of Core-Periphery
JD	42.2007	Networks
SD	43.2009	Fabio Eboli, Ramiro Parrado, Roberto Roson: <u>Climate Change Feedback on Economic Growth: Explorations</u>
-		with a Dynamic General Equilibrium Mode
GC	44.2009	Fabio Sabatini: Does Social Capital Create Trust? Evidence from a Community of Entrepreneurs
SD	45.2009	ZhongXiang Zhang: Is it Fair to Treat China as a Christmas Tree to Hang Everybody's Complaints? Putting
		its Own Energy Saving into Perspective
SD	46.2009	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Emmanuel Petrakis: The Role of Information Provision
		as a Policy Instrument to Supplement Environmental Taxes: Empowering Consumers to Choose Optimally
SD	47.2009	Jean-François Caulier, Ana Mauleon and Vincent Vannetelbosch: Contractually Stable Networks
GC	48.2009	Massimiliano Mazzanti, Susanna Mancinelli, Giovanni Ponti and Nora Piva: Education, Reputation or
		Network? Evidence from Italy on Migrant Workers Employability
SD	49.2009	William Brock and Anastasios Xepapadeas: General Pattern Formation in Recursive Dynamical Systems
		Models in Economics
SD	50.2009	Giovanni Marin and Massimiliano Mazzanti: Emissions Trends and Labour Productivity Dynamics Sector
CD	54 2000	Analyses of De-coupling/Recoupling on a 1990-2005 Namea
SD	51.2009	Yoshio Kamijo and Ryo Kawasaki (lxxxv): <u>Dynamics, Stability, and Foresight in the Shapley-Scarf Housing</u>
18.4	52 2000	Market
IM	52.2009	Laura Poddi and Sergio Vergalli: <u>Does Corporate Social Responsibility Affect the Performance of Firms?</u>
SD	53.2009	Valentina Bosetti, Carlo Carraro and Massimo Tavoni: <u>Climate Change Mitigation Strategies in Fast-</u> Growing Countries: The Benefits of Early Action
GC	54.2009	Alireza Naghavi and Gianmarco I.P. Ottaviano: <u>Firm Heterogeneity</u> , <u>Contract Enforcement</u> , and the Industry
uc	34.2009	Dynamics of Offshoring
IM	55.2009	Giacomo Calzolari and Carlo Scarpa: On Regulation and Competition: Pros and Cons of a Diversified
	00.2005	Monopolist
SD	56.2009	Valentina Bosetti, Ruben Lubowski and Alexander Golub and Anil Markandya: <u>Linking Reduced</u>
		Deforestation and a Global Carbon Market: Impacts on Costs, Financial Flows, and Technological
		Innovation
IM	57.2009	Emmanuel Farhi and Jean Tirole: Collective Moral Hazard, Maturity Mismatch and Systemic Bailouts
SD	58.2009	Kelly C. de Bruin and Rob B. Dellink: How Harmful are Adaptation Restrictions
SD	59.2009	Rob Dellink, Michel den Elzen, Harry Aiking, Emmy Bergsma, Frans Berkhout, Thijs Dekker, Joyeeta Gupta:
		Sharing the Burden of Adaptation Financing: An Assessment of the Contributions of Countries
SD	60.2009	Stefania Tonin, Anna Alberini and Margherita Turvani: The Value of Reducing Cancer Risks at Contaminated
		Sites: Are More Heavily Exposed People Willing to Pay More?
SD	61.2009	Clara Costa Duarte, Maria A. Cunha-e-Sá and Renato Rosa: <u>The Role of Forests as Carbon Sinks: Land-Use</u>
		and Carbon Accounting
GC	62.2009	Carlo Altomonte and Gabor Békés: Trade Complexity and Productivity
GC	63.2009	Elena Bellini, Gianmarco I.P. Ottaviano, Dino Pinelli and Giovanni Prarolo: <u>Cultural Diversity and Economic</u>
CD	64 2000	<u>Performance: Evidence from European Regions</u> Valentina Bosetti, Carlo Carraro, Enrica De Cian, Romain Duval, Emanuele Massetti and Massimo Tavoni:
SD	64.2009	The Incentives to Participate in, and the Stability of, International Climate Coalitions: A Game-theoretic
		Analysis Using the Witch Model
IM	65.2009	John Temple Lang: Article 82 EC - The Problems and The Solution
SD	66.2009	P. Dumas and S. Hallegatte: Think Again: Higher Elasticity of Substitution Increases Economic Resilience
SD	67.2009	Ruslana Rachel Palatnik and Roberto Roson: Climate Change Assessment and Agriculture in General
JD	07.2007	Equilibrium Models: Alternative Modeling Strategies
SD		
	68.2009	Paulo A.L.D. Nunes, Helen Ding and Anil Markandva: The Economic Valuation of Marine Ecosystems
IM	68.2009 69.2009	Paulo A.L.D. Nunes, Helen Ding and Anil Markandya: <u>The Economic Valuation of Marine Ecosystems</u> Andreas Madestam: Informal Finance: A Theory of Moneylenders
IM SD	69.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders
IM SD		Andreas Madestam: <u>Informal Finance: A Theory of Moneylenders</u> Efthymia Kyriakopoulou and Anastasios Xepapadeas: <u>Environmental Policy, Spatial Spillovers and the</u>
	69.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations
SD	69.2009 70.2009	Andreas Madestam: <u>Informal Finance: A Theory of Moneylenders</u> Efthymia Kyriakopoulou and Anastasios Xepapadeas: <u>Environmental Policy, Spatial Spillovers and the</u>
SD	69.2009 70.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean:
SD SD	69.2009 70.2009 71.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives
SD SD	69.2009 70.2009 71.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan
SD SD GC	69.2009 70.2009 71.2009 72.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan Cities
SD SD GC SD GC	69.2009 70.2009 71.2009 72.2009 73.2009 74.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan Cities  Erik Ansink: Self-enforcing Agreements on Water allocation  Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: Mapping the Evolution of "Clusters": A Meta-analysis
SD SD GC SD GC SD	69.2009 70.2009 71.2009 72.2009 73.2009 74.2009 75.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan Cities  Erik Ansink: Self-enforcing Agreements on Water allocation  Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: Mapping the Evolution of "Clusters": A Meta-analysis  Nektarios Aslanidis: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey
SD SD GC SD GC SD SD SD	69.2009 70.2009 71.2009 72.2009 73.2009 74.2009 75.2009 76.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan Cities  Erik Ansink: Self-enforcing Agreements on Water allocation  Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: Mapping the Evolution of "Clusters": A Meta-analysis  Nektarios Aslanidis: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey  Joan Canton: Environmentalists' Behaviour and Environmental Policies
SD SD GC SD GC SD	69.2009 70.2009 71.2009 72.2009 73.2009 74.2009 75.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders  Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the Emergence of Economic Agglomerations  A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: Coastal Zone Management in the Mediterranean: Legal and Economic Perspectives  Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan Cities  Erik Ansink: Self-enforcing Agreements on Water allocation  Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: Mapping the Evolution of "Clusters": A Meta-analysis  Nektarios Aslanidis: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey

IM	78.2009	Chiara D'Alpaos, Michele Moretto, Paola Valbonesi and Sergio Vergalli: "It Is Never too late": Optimal
		Penalty for Investment Delay in Public Procurement Contracts
SD	79.2009	Henry Tulkens and Vincent van Steenberghe: "Mitigation, Adaptation, Suffering": In Search of the Right Mix
		in the Face of Climate Change
SD	80.2009	Giovanni Bella: A Search Model for Joint Implementation
SD	81.2009	ZhongXiang Zhang: Multilateral Trade Measures in a Post-2012 Climate Change Regime?: What Can Be
		Taken from the Montreal Protocol and the WTO?
SD	82.2009	Antoine Dechezleprêtre, Matthieu Glachant, Ivan Hascic, Nick Johnstone and Yann Ménière: Invention and
		<u>Transfer of Climate Change Mitigation Technologies on a Global Scale: A Study Drawing on Patent Data</u>
SD	83.2009	László Á. Kóczy: Stationary Consistent Equilibrium Coalition Structures Constitute the Recursive Core
SD	84.2009	Luca Di Corato and Michele Moretto: Investing in Biogas: Timing, Technological Choice and the Value of
		Flexibility from Inputs Mix
SD	85.2009	Valentina Bosetti, Enrica De Cian, Alessandra Sgobbi, and Massimo Tavoni: <u>The 2008 WITCH Model: New</u>
		Model Features and Baseline
IM	86.2009	Rocco Macchiavello: Vertical Integration and Investor Protection in Developing Countries
SD	87.2009	Massimiliano Mazzanti and Antonio Musolesi: Carbon Kuznets Curves: Long-run Structural Dynamics and
		Policy Events
GC	88.2009	Gianmarco I.P. Ottaviano and Christian Volpe Martincus: SMEs in Argentina: Who are the Exporters
GC	89.2009	Gianpaolo Rossini and Cecilia Vergari: Input Production Joint Venture
SD	90.2009	Angelo Antoci, Simone Borghesi and Marcello Galeotti: Environmental Options and Technological
		Innovation: An Evolutionary Game Model
GC	91.2009	Cristina Cattaneo: The Decision to Migrate and Social Capital: Evidence from Albania

(lxxxv) This paper has been presented at the 14th Coalition Theory Network Workshop held in Maastricht, The Netherlands, on 23-24 January 2009 and organised by the Maastricht University CTN group (Department of Economics, http://www.feem-web.it/ctn/12d\_maa.php).